# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# FORM S-8 REGISTRATION STATEMENT

UNDER THE SECURITIES ACT OF 1933

# LANTHEUS HOLDINGS, INC.

(Exact name of registrant as specified in its charter)

Delaware (State or Other Jurisdiction of Incorporation) 35-2318913 (IRS Employer Identification No.)

331 Treble Cove Road, North Billerica, Massachusetts (Address of principal executive offices)

01862 (Zip Code)

Lantheus Holdings, Inc. 2005 Stock Incentive Plan Lantheus Holdings, Inc. 2018 Performance Incentive Plan (Full title of the plan)

Michael P. Duffy
Senior Vice President and General Counsel
331 Treble Cove Road, Building 600-2
North Billerica, Massachusetts 01862
(Name and address of agent for service)

Telephone: (978) 671-8408 (Telephone number, including area code, of agent for service)

Copies of communications, including communications sent to agent for service to:

Morton A. Pierce, Esq. Bryan J. Luchs, Esq. Michelle B. Rutta, Esq. White & Case LLP 1221 Avenue of the Americas New York, New York 10020 (212) 819-8200

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.

company in rune 120 2 0	The Exchange rect				
Large accelerated filer	$\boxtimes$	Accelerated filer			
Non-accelerated filer		Smaller reporting company			
		Emerging growth company			
If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section $7(a)(2)(B)$ of the Securities Act. $\Box$					

# CALCULATION OF REGISTRATION FEE

Title of Securities to be Registered	Amount to be Registered(1)	Proposed Maximum Offering Price Per Share(2)	Proposed Maximum Aggregate Offering Price	Amount of Registration Fee
Common Stock, par value \$0.01 per share	4,211,290	\$ 13.44	\$ 56,599,737.60	\$ 7,346.65

- (1) Represents shares of Lantheus Holdings, Inc. (the "Registrant") common stock, par value \$0.01 per share (the "Common Stock") that may be issued pursuant to the Lantheus Holdings, Inc. 2018 Performance Incentive Plan (f/k/a Progenics Pharmaceuticals, Inc. 2018 Performance Incentive Plan), as amended, and the Lantheus Holdings, Inc. 2005 Stock Incentive Plan (f/k/a Progenics Pharmaceuticals, Inc. 2005 Stock Incentive Plan), as amended (together, the "Plans"). In addition, pursuant to Rule 416(a) under the Securities Act of 1933, as amended (the "Securities Act"), this Registration Statement on Form S-8 (this "Registration Statement") also covers an indeterminable number of additional securities that may become issuable under the Plans, as amended, as a result of any stock splits, stock dividends, recapitalizations or similar transaction.
- (2) Estimated solely for purposes of calculating the registration fee pursuant to Rule 457(c) and Rule 457(h) under the Securities Act, based on the average of the high and low sales prices of the Registrant's common stock on The NASDAQ Global Market on June 24, 2019.

#### **EXPLANATORY NOTE**

This Registration Statement is being filed for the purpose of registering 4,211,290 shares of Common Stock that may be issued pursuant to the Plans in connection with the merger ("Merger") of Plato Merger Sub, Inc., a Delaware corporation and wholly-owned subsidiary of the Registrant ("Merger Sub"), with and into Progenics Pharmaceuticals, Inc. ("Progenics"), pursuant to that certain Amended and Restated Agreement and Plan of Merger, dated as of February 20, 2020 (the "Merger Agreement"), by and among the Registrant, Merger Sub and Progenics. On June 19, 2020, upon effectiveness of the Merger (the "Effective Time"), Progenics became a wholly-owned subsidiary of the Registrant.

Pursuant to the Merger Agreement, at the Effective Time, each in-the-money Progenics stock option was assumed by the Registrant and converted into an option to purchase (a) that number of shares of Common Stock (rounded down to the nearest whole share) equal to the product obtained by multiplying (i) the total number of shares of Progenics common stock subject to such Progenics stock option immediately prior to the Effective Time by (ii) 0.31, (b) at a per-share exercise price (rounded up to the nearest whole cent) equal to the quotient obtained by dividing (i) the exercise price per share of Progenics common stock at which such Progenics stock option was exercisable immediately prior to the Effective Time by (ii) 0.31. Each such in-the-money assumed stock option continued to have the same terms and conditions (including the applicable time-vesting and/or performance-vesting conditions and any provisions for accelerated vesting) as applied to the corresponding Progenics in-the-money stock option immediately prior to the Effective Time. Each holder of a Progenics in-the-money stock option also received a vested or unvested contingent value right depending on whether the underlying in-the-money Progenics stock option was vested at the Effective Time.

In addition, at the Effective Time, each out-of-the-money Progenics stock option was assumed by the Registrant and converted into an option to purchase (a) that number of shares of Common Stock (rounded down to the nearest whole share) equal to the product obtained by multiplying (i) the total number of shares of Progenics common stock that were subject to such out-of-the-money stock option immediately prior to the Effective Time by (ii) 0.31, (b) at a per-share exercise price (rounded up to the nearest whole cent) equal to the quotient obtained by dividing (i) the exercise price per share of Progenics common stock at which such Progenics stock option was exercisable immediately prior to the Effective Time by (ii) 0.31.

#### PART I

# INFORMATION REQUIRED IN THE SECTION 10(A) PROSPECTUS

The information required by Item 1 and Item 2 of Part I of Form S-8 is omitted from this filing in accordance with Rule 428 under the Securities Act and the introductory note to Part I of Form S-8. The documents containing the information specified in Part I will be delivered to the participants in the plan covered by this Registration Statement as required by Rule 428(b)(1).

#### PART II

# INFORMATION REQUIRED IN THE REGISTRATION STATEMENT

# Item 3. Incorporation of Documents by Reference.

The following documents filed with the Securities and Exchange Commission (the "<u>SEC</u>") by the Registrant pursuant to the Securities Act and the Securities Exchange Act of 1934, as amended (the "<u>Exchange Act</u>") are incorporated herein by reference:

- (a) the Registrant's Annual Report on Form 10-K for the fiscal year ended December 31, 2019, filed with the SEC on February 25, 2020;
- (b) the information specifically incorporated by reference into the Registrant's Annual Report on Form 10-K for the fiscal year ended December 31, 2019 from the Registrant's Definitive Proxy Statement on Schedule 14A filed with the SEC on March 12, 2020;
- (c) the Registrant's Quarterly Report on Form 10-Q for the quarterly period ended March 31, 2020, filed with the SEC on April 30, 2020;
- (d) the Registrant's Current Reports on Form 8-K, filed with the SEC on <u>January 13, 2020</u>, <u>January 27, 2020</u>, <u>February 3, 2020</u>, <u>February 20, 2020</u>, <u>April 2, 2020</u>, <u>April 10, 2020</u>, <u>April 14, 2020</u>, <u>April 24, 2020</u>, <u>June 16, 2020</u> and <u>June 22, 2020</u>; and
- (e) the description of Common Stock, which is contained in the Registrant's registration statement on Form S-4 filed with the SEC on November 12, 2019, including any amendments, supplement or report filed for the purpose of updating such description.

All reports and other documents filed by the Registrant after the date hereof pursuant to Sections 13(a), 13(c), 14 and 15(d) of the Exchange Act, prior to the filing of a post-effective amendment to this Registration Statement that indicates that all securities offered have been sold or which deregisters all securities then remaining unsold, shall be deemed to be incorporated herein by reference and to be part hereof from the date of filing of such documents. Any statement contained herein or in a document incorporated or deemed to be incorporated by reference herein shall be deemed to be modified or superseded for purposes of this Registration Statement to the extent that a statement contained herein or in any subsequently filed document which also is incorporated or is deemed to be incorporated by reference herein modifies or supersedes such earlier statement. Any statement so modified or superseded shall not be deemed, except as so modified or superseded, to constitute part of this Registration Statement.

# Item 4. Description of Securities.

Not applicable.

## Item 5. Interests of Named Experts and Counsel.

Not applicable.

## Item 6. Indemnification of Directors and Officers.

Section 145 of the Delaware General Corporation Law (the "<u>DGCL</u>") provides that a corporation may indemnify directors and officers as well as other employees and individuals against expenses (including attorneys' fees), judgments, fines and amounts paid in settlement in connection with specified actions, suits or proceedings, whether civil, criminal, administrative or investigative (other than an action by or in the right of the corporation, a "derivative action"), if they acted in good faith and in a manner they reasonably believed to be in or not opposed to the best interests of the corporation and, with respect to any criminal actions or proceedings, had no reasonable cause to believe their conduct was unlawful. A similar standard is applicable

in the case of derivative actions, except that indemnification only extends to expenses (including attorneys' fees) actually and reasonably incurred in connection with the defense or settlement of such action, and the DGCL requires court approval before there can be any indemnification where the person seeking indemnification has been found liable to the corporation. The DGCL provides that it is not exclusive of other indemnification that may be granted by a corporation's by-laws, disinterested director vote, stockholder vote, agreement or otherwise.

Under the terms of the Registrant's by-laws and subject to the applicable provisions of Delaware law, the Registrant has agreed to indemnify its directors and officers and, subject to the discretion of the board of directors, any employee or agent of the Registrant, against expenses actually and reasonably incurred in connection with any actual, threatened or completed proceeding, in which such director or officer may be involved by reason of being or having been a director or officer of the Registrant, or of serving or having served at the Registrant's request as a director, officer, trustee, general partner, managing member, fiduciary, employee or agent of any other corporation, constituent corporation (including any constituent of a constituent) absorbed in a consolidation or merger to which the Registrant (or any of their wholly-owned subsidiaries) is a party, limited liability company, partnership, joint venture, trust, employee benefit plan or other enterprise.

Section 102(b)(7) of the DGCL permits a provision in the certificate of incorporation of each corporation organized thereunder, such as the Registrant, eliminating or limiting, with certain exceptions, the personal liability of a director to the corporation or its stockholders for monetary damages for breach of fiduciary duty as a director. The Registrant's certificate of incorporation eliminates the liability of directors to the extent permitted by the DGCL.

The Registrant carries directors' and officers' liability insurance that covers certain liabilities and expenses of its directors and officers.

# Item 7. Exemption from Registration Claimed.

Not applicable.

#### Item 8. Exhibits.

Exhibit No.	<u>Description</u>
4.1	Amended and Restated Certificate of Incorporation of Lantheus Holdings, Inc. (incorporated herein by reference to Exhibit 3.1 to the Current Report on Form 8-K of Lantheus Holdings, Inc. dated April 26, 2018 and filed April 27, 2018).
4.2	Amended and Restated Bylaws of Lantheus Holdings, Inc. (incorporated herein by reference to Exhibit 3.2 to the Current Report on Form 8-K of Lantheus Holdings, Inc. dated April 26, 2018 and filed April 27, 2018).
4.3	Specimen Common Stock Certificate of Lantheus Holdings, Inc. (incorporated herein by reference to Exhibit 4.1 to the Current Report on Form 8-K of Lantheus Holdings, Inc., dated June 30, 2015).
4.4	<u>Lantheus Holdings, Inc. 2005 Stock Incentive Plan (f/k/a Progenics Pharmaceuticals, Inc. 2005 Stock Incentive Plan).*</u>
4.5	<u>Lantheus Holdings, Inc. 2018 Performance Incentive Plan (f/k/a Progenics Pharmaceuticals, Inc. 2018 Performance Incentive Plan).*</u>
5.1	Opinion of White & Case LLP regarding the validity of the securities being issued.*
23.1	Consent of Deloitte & Touche LLP Independent Registered Public Accounting Firm of Lantheus Holdings, Inc.*
23.2	Consent of Ernst & Young LLP Independent Registered Public Accounting Firm of Progenics Pharmaceuticals, Inc.*
23.3	Consent of White & Case LLP (included as part of its opinion filed as Exhibit 5.1).*
24.1	Power of Attorney (included in the signature page of this Form S-8).*

<sup>\*</sup> Filed herewith

# Item 9. Undertakings.

- (a) The undersigned Registrant hereby undertakes:
  - (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this registration statement:
    - (i) To include any prospectus required by Section 10(a)(3) of the Securities Act;
    - (ii) To reflect in the prospectus any facts or events arising after the effective date of this registration statement (or the most recent post-effective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in this Registration Statement. Notwithstanding the foregoing, any increase or decrease in volume of securities offered (if the total dollar value of securities offered would not exceed that which was registered) and any deviation from the low or high end of the estimated maximum offering range may be reflected in the form of prospectus filed with the Commission pursuant to Rule 424(b) (§ 230.424(b) of this chapter) if, in the aggregate, the changes in volume and price represent no more than 20% change in the maximum aggregate offering price set forth in the "Calculation of Registration Fee" table in the effective registration statement; and
    - (iii) To include any material information with respect to the plan of distribution not previously disclosed in this Registration Statement or any material change to such information in this Registration Statement;

*provided*, *however*, that paragraphs (a)(1)(i) and (a)(1)(ii) do not apply if the information required to be included in a post-effective amendment by those paragraphs is contained in reports filed with or furnished to the SEC by the Registrant pursuant to Section 13 or Section 15(d) of the Securities Exchange Act (15 U.S.C. 78m or 78o(d)) that are incorporated by reference in this Registration Statement.

- (2) That, for the purpose of determining any liability under the Securities Act, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.
- (b) The undersigned Registrant hereby undertakes that, for purposes of determining any liability under the Securities Act, each filing of the Registrant's annual report pursuant to Section 13(a) or Section 15(d) of the Exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the Exchange Act) that is incorporated by reference in this Registration Statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (c) Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers and controlling persons of the Registrant pursuant to the foregoing provisions, or otherwise, the Registrant has been advised that in the opinion of the SEC such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the Registrant of expenses incurred or paid by a director, officer or controlling person of the Registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the Registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act and will be governed by the final adjudication of such issue.

#### **SIGNATURES**

Pursuant to the requirements of the Securities Act of 1933, the registrant certifies that it has reasonable grounds to believe that it meets all of the requirements for filing on Form S-8 and has duly caused this registration statement to be signed on its behalf by the undersigned, hereunto duly authorized, on this 26th day of June, 2020.

# LANTHEUS HOLDINGS, INC.

By: /s/ Mary Anne Heino

Name: Mary Anne Heino

Title: President and Chief Executive Officer

#### POWER OF ATTORNEY

KNOW ALL PERSONS BY THESE PRESENTS, that each of the undersigned constitutes and appoints each of Mary Anne Heino, Robert J. Marshall, Jr. and Michael P. Duffy, each acting alone, her or his true and lawful attorney-in-fact and agent, with full power of substitution and resubstitution, for such person and in her or his name, place and stead, in any and all capacities, to sign this Registration Statement on Form S-8 (including all pre-effective and post-effective amendments and registration statements filed pursuant to Rule 462(b) under the Securities Act of 1933), and to file the same, with all exhibits thereto, and other documents in connection therewith, with the Securities and Exchange Commission, granting unto said attorneys-in-fact and agents, each acting alone, full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises, as fully to all intents and purposes as he might or could do in person, hereby ratifying and confirming that any such attorney-in-fact and agent, or her or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act of 1933, this Registration Statement has been signed by the following persons in the capacities and on the dates indicated.

Signature	Title	Date
/s/ Mary Anne Heino Mary Anne Heino	President, Chief Executive Officer and Director (Principal Executive Officer)	June 26, 2020
/s/ Robert J. Marshall, Jr. Robert J. Marshall, Jr.	Chief Financial Officer (Principal Financial and Accounting Officer)	June 26, 2020
/s/ Brian Markison Brian Markison	Chairman of the Board	June 26, 2020
/s/ Dr. Gérard Ber Dr. Gérard Ber	Director	June 26, 2020
/s/ Samuel Leno Samuel Leno	Director	June 26, 2020
/s/ Heinz Mäusli Heinz Mäusli	Director	June 26, 2020
/s/ Julie H. McHugh Julie H. McHugh	Director	June 26, 2020
/s/ Gary Pruden Gary Pruden	Director	June 26, 2020
/s/ Dr. Frederick Robertson Dr. Frederick Robertson	Director	June 26, 2020
/s/ Dr. James Thrall Dr. James Thrall	Director	June 26, 2020

# LANTHEUS HOLDINGS, INC. 2005 STOCK INCENTIVE PLAN (F/K/A PROGENICS PHARMACEUTICALS, INC. 2005 STOCK INCENTIVE PLAN)

(As amended, effective June 19, 2020)

#### 1. PURPOSE

The purpose of the Lantheus Holdings, Inc. 2005 Stock Incentive Plan (formerly known as the Progenics Pharmaceuticals, Inc. 2005 Stock Incentive Plan) is to further align the interests of employees, officers, non-employee directors and other individual service providers with those of the stockholders by providing incentive compensation opportunities tied to the performance of the Common Stock and by promoting increased ownership of the Common Stock by such individuals. The Plan is also intended to advance the interests of the Company and its stockholders by attracting, retaining and motivating key personnel upon whose judgment, initiative and effort the successful conduct of the Company's business is largely dependent.

#### 2. DEFINITIONS

Wherever the following capitalized terms are used in the Plan, they shall have the meanings specified below:

- (a) Award means an award of a Stock Option, Stock Appreciation Right, Restricted Stock Award, Stock Unit Award, Stock Award or Performance Award granted under the Plan.
- **(b) Award Agreement** means a written or electronic agreement entered into between the Company and a Participant setting forth the terms and conditions of an Award.
- **(c) Board** means the Board of Directors of the Company.
- **(d) Change in Control** shall have the meaning set forth in Section 13.2 hereof.
- (e) Code means the Internal Revenue Code of 1986, as amended.
- **(f) Committee** means the Compensation Committee of the Board or a successor thereof, or any other committee of the Board appointed by the Board to administer the Plan from time to time.
- **(g) Common Stock** means the Company's Common Stock, par value \$.01 per share.
- (h) Company means Lantheus Holdings, Inc., a Delaware corporation.
- (i) Date of Grant means the date on which an Award under the Plan is granted by the Committee, or such later date as the Committee may specify to be the effective date of an Award.
- (j) **Disability** means a Participant being considered "disabled" within the meaning of Section 409A(a)(2)(C) of the Code, unless otherwise provided in an Award Agreement.
- **(k) Eligible Person** means any person who is an employee, officer, director, consultant, advisor or other individual service provider of the Company or any Subsidiary, as determined by the Committee, or any person who is determined by the Committee to be a prospective employee, officer, director, consultant, advisor or other individual service provider of the Company or any Subsidiary.
- (l) Exchange Act means the Securities Exchange Act of 1934, as amended.

- (m) Fair Market Value with respect to the value of a share of Common Stock as of a particular day, shall mean the last reported sale price (as reported on the NASDAQ) of the Common Stock on such day (unless such day is not a trading day, in which case, on the last trading day immediately preceding such day on which the Common Stock is traded on the NASDAQ). If the Common Stock is not listed on the NASDAQ, the Committee shall determine in good faith the Fair Market Value in whatever manner it considers appropriate, taking into account to the extent necessary the requirements of Section 409A of the Code.
- (n) Incentive Stock Option means a Stock Option granted under Section 6 hereof that is intended to meet the requirements of section 422 of the Code and the regulations promulgated thereunder.
- **(o) Merger** means the acquisition of all outstanding shares of common stock, par value \$0.0013 per share, of Progenics by the Company by way of a merger of Plato Merger Sub, Inc., a Delaware corporation and a wholly-owned subsidiary of the Company, with and into Progenics, with Progenics being the surviving corporation.
- (p) NASDAQ means The NASDAQ Stock Market's National Market.
- (q) Nonqualified Stock Option means a Stock Option granted under Section 6 hereof that is not an Incentive Stock Option.
- (r) Participant means any Eligible Person who holds an outstanding Award under the Plan.
- (s) **Performance Awards** means an Award under Section 11 hereof entitling a Participant to a payment in cash at the end of a performance period, if the performance and other conditions established by the Committee are satisfied.
- (t) Plan means this Lantheus Holdings, Inc. 2005 Stock Incentive Plan as amended herein, and as may be amended from time to time, effective as provided in Section 15.1 hereof.
- (u) **Progenics** means Progenics Pharmaceuticals, Inc.
- (v) **Progenics Board** means the Board of Directors of Progenics.
- (w) Restricted Stock Award means a grant of shares of Common Stock to an Eligible Person under Section 8 hereof that are issued subject to such vesting and transfer restrictions and such other conditions as are set forth in the Plan and the applicable Award Agreement.
- (x) Section 162(m) Award means any Award that is intended to qualify for the "performance-based" compensation exception under section 162(m) of the Code and the regulations promulgated thereunder.
- (y) Service means a Participant's employment or other service relationship with Progenics, the Company or any Subsidiary.
- (z) Stock Appreciation Right means a contractual right granted to an Eligible Person under Section 7 hereof entitling such Eligible Person to receive a payment, representing the difference between the base price per share of the right and the Fair Market Value of a share of Common Stock at such time, and subject to such conditions, as are set forth in the Plan and the applicable Award Agreement.
- **(aa) Stock Award** means a grant of shares of Common Stock to an Eligible Person under Section 10 hereof entitling a Participant to shares of Common Stock that are issued free of transfer restrictions and forfeiture conditions.

- **(bb) Stock Option** means a contractual right granted to an Eligible Person under Section 6 hereof to purchase shares of Common Stock at such time and price, and subject to such conditions, as are set forth in the Plan and the applicable Award Agreement.
- (cc) Stock Unit Award means a contractual right granted to an Eligible Person under Section 9 hereof representing notional unit interests equal in value to a share of Common Stock to be paid and distributed at such times, and subject to such conditions, as are set forth in the Plan and the applicable Award Agreement.
- **(dd) Subsidiary** means an entity (whether or not a corporation) that is wholly or majority owned or controlled, directly or indirectly, by the Company, or any other affiliate of the Company that is so designated, from time to time, by the Committee; provided, however, that with respect to Incentive Stock Options, the term "Subsidiary" shall include only an entity that qualifies under section 424(f) of the Code as a "subsidiary corporation" with respect to the Company.

#### 3. ADMINISTRATION

Section 3.1 Committee Members. The Plan shall be administered by a Committee comprised of no fewer than two members of the Board. It is intended that each Committee member shall satisfy the requirements for (i) an "independent director" under rules adopted by the NASDAQ, (ii) a "nonemployee director" for purposes of such Rule 16b-3 under the Exchange Act and (iii) to the extent applicable, an "outside director" under section 162(m) of the Code. No member of the Committee shall be liable for any action or determination made in good faith by the Committee with respect to the Plan or any Award hereunder.

Section 3.2 Committee Authority. The Committee shall have such powers and authority as may be necessary or appropriate for the Committee to carry out its functions as described in the Plan. Subject to the express limitations of the Plan, the Committee shall have authority in its discretion to determine the Eligible Persons to whom, and the time or times at which, Awards may be granted, the number of shares, units or other rights subject to each Award, the exercise, base or purchase price of an Award (if any), the time or times at which an Award will become vested, exercisable or payable, the performance criteria, performance goals and other conditions of an Award, the duration of the Award, and all other terms of the Award. Subject to the terms of the Plan, the Committee shall have the authority to amend the terms of an Award in any manner that is not inconsistent with the Plan, provided that no such action shall adversely affect the rights of a Participant with respect to an outstanding Award without the Participant's consent. The Committee shall also have discretionary authority to interpret the Plan, to make all factual determinations under the Plan, and to make all other determinations necessary or advisable for Plan administration, including, without limitation, to correct any defect, to supply any omission or to reconcile any inconsistency in the Plan or any Award Agreement hereunder. The Committee may prescribe, amend, and rescind rules and regulations relating to the Plan. The Committee's determinations under the Plan need not be uniform and may be made by the Committee selectively among Participants and Eligible Persons, whether or not such persons are similarly situated. The Committee shall, in its discretion, consider such factors as it deems relevant in making its interpretations, determinations and actions under the Plan including, without limitation, the recommendations or advice of any officer or employee of the Company or such attorneys, consultants, accountants or other advisors as it may select. All interpretations, determinations, and actions by the Committee shall be final, conclusive, and binding upon all parties. Notwithstanding anything to the contrary contained herein, the Board may, in its sole discretion, at any time and from time to time, exercise all of the authority and responsibility granted to the Committee herein, including the granting or amending of Awards.

Section 3.3 Delegation of Authority. The Committee shall have the right, from time to time, to delegate to one or more officers of the Company the authority of the Committee to grant Awards under the Plan, subject to the requirements of section 157(c) of the Delaware General Corporation Law (or any successor provision) and such other limitations as the Committee shall determine. In no event shall any such delegation of authority be permitted with respect to Awards to any members of the Board or to any Eligible Person who is subject to Rule 16b-3 under the Exchange Act or who is a covered employee under section 162(m) of the Code. The Committee shall also be permitted to delegate, to any appropriate officer or employee of the Company, responsibility for performing ministerial functions under the Plan. In the event that the Committee's authority is delegated to officers or employees in accordance with the foregoing, all provisions of the Plan relating to the Committee shall be interpreted in a manner consistent with the foregoing by treating any such reference as a reference to such officer or employee for such purpose. Any action undertaken in accordance with the Committee's delegation of authority hereunder shall have the same force and effect as if such action was undertaken directly by the Committee and shall be deemed for all purposes of the Plan to have been taken by the Committee.

**Section 3.4 Grants to Non-Employee Directors.** Any Awards or formula for granting Awards under the Plan made to non-employee directors shall be approved by the Board. With respect to awards granted to such directors, all rights, powers and authorities vested in the Committee under the Plan shall instead be exercised by the Board, and all provisions of the Plan relating to the Committee shall be interpreted in a manner consistent with the foregoing by treating any such reference as a reference to the Board for such purpose.

#### 4. SHARES SUBJECT TO THE PLAN

Section 4.1 Share Limitation. Subject to adjustment pursuant to Section 4.2 hereof, the maximum aggregate number of shares of Common Stock which may be issued under all Awards granted to Participants under the Plan shall be 3,549,500 shares (this being the adjusted number upon assumption of the Plan in connection with the Merger). Shares of Common Stock issued under the Plan may be either authorized but unissued shares or shares held in the Company's treasury. Shares of Common Stock subject to Awards of Stock Options or Stock Appreciation Rights that are settled in Common Stock shall be counted against the maximum share limitations of this Section 4.1 as one share of Common Stock for every share of Common Stock subject thereto, regardless of the number of shares of Common Stock actually issued to settle the Stock Option or Stock Appreciation Right upon exercise. Shares of Common Stock subject to Awards of Restricted Stock Awards, Stock Unit Awards, Stock Awards, or share-based Performance Awards, if any, shall be counted against the maximum share limitations of this Section 4.1 as 1.2 shares of Common Stock for every share of Common Stock subject thereto. To the extent that any Award under the Plan payable in shares of Common Stock is forfeited, cancelled, returned to the Company for failure to satisfy vesting requirements or upon the occurrence of other forfeiture events, or otherwise terminates, in whole or in part, without payment being made thereunder, the shares of Common Stock remaining subject thereto at the time of such forfeiture, cancellation, return or other termination will no longer be counted against the foregoing maximum share limitations and may again be made subject to Awards under the Plan pursuant to such limitations. Awards under the Plan that are settled in cash and not in shares of Common Stock shall not be counted against the foregoing maximum share limitations. Shares that are withheld from an Award by the Participant in payment of the exercise or purchase price or separately surrendered

Section 4.2 Adjustments. If there shall occur any change with respect to the outstanding shares of Common Stock by reason of any recapitalization, reclassification, stock dividend, extraordinary dividend, stock split, reverse stock split, or other distribution with respect to the shares of Common Stock, or any merger, reorganization, consolidation, combination, spin-off or other similar corporate change, or any other change affecting the Common Stock, the Committee shall, in the manner and to the extent that it deems appropriate and equitable to the Participants and consistent with the terms of the Plan, cause an adjustment to be made in (i) the maximum numbers and kind of shares provided in Section 4.1 hereof, (ii) the maximum

numbers and kind of shares or units set forth in Sections 6.1, 7.1, 8.1, 9.1 and 10.1 hereof, (iii) the numbers and kind of shares of Common Stock, units, or other rights subject to then outstanding Awards, (iv) the price for each share or unit or other right subject to then outstanding Awards, (v) the performance measures or goals relating to an Award and (vi) any other terms of an Award that are affected by the event to prevent dilution or enlargement of a Participant's rights under an Award. Notwithstanding the foregoing, in the case of Incentive Stock Options, any such adjustments shall, to the extent practicable, be made in a manner consistent with the requirements of section 424(a) of the Code.

# 5. PARTICIPATION AND AWARDS

Section 5.1 Designation of Participants. No new Awards will be granted under the Plan following the Merger.

# 6. STOCK OPTIONS

**Section 6.1 Grant of Stock Option.** A Stock Option may be granted to any Eligible Person selected by the Committee. Subject to the provisions of Section 6.7 hereof and section 422 of the Code, each Stock Option shall be designated, in the discretion of the Committee, as an Incentive Stock Option or as a Nonqualified Stock Option. The maximum number of shares of Common Stock that may be subject to Stock Options granted to any Participant during any calendar year shall be limited to 750,000 shares (subject to adjustment as provided in Section 4.2 hereof).

**Section 6.2 Exercise Price**. The exercise price per share of a Stock Option shall not be less than 100 percent of the Fair Market Value of the shares of Common Stock on the Date of Grant, provided that the Committee may in its discretion specify for any Stock Option an exercise price per share that is higher than the Fair Market Value on the Date of Grant.

**Section 6.3 Vesting of Stock Options**. The Committee shall in its discretion prescribe the time or times at which, or the conditions upon which, a Stock Option or portion thereof shall become vested and/or exercisable. The requirements for vesting and exercisability of a Stock Option may be based on the continued Service of the Participant with the Company or a Subsidiary for a specified time period (or periods) or on the attainment of a specified performance goal (or goals) established by the Committee in its discretion. The Committee may, in its discretion, accelerate the vesting or exercisability of any Stock Option at any time.

Section 6.4 Term of Stock Options. The Committee shall in its discretion prescribe in an Award Agreement the period during which a vested Stock Option may be exercised, provided that the maximum term of a Stock Option shall be ten years from the Date of Grant. A Stock Option may be earlier terminated as specified by the Committee and set forth in an Award Agreement upon or following the termination of a Participant's Service with the Company or any Subsidiary, including by reason of voluntary resignation, death, Disability, termination for cause or any other reason. Except as otherwise provided in this Section 6 or in an Award Agreement, no Stock Option may be exercised at any time during the term thereof unless the Participant is then in the Service of the Company or one of its Subsidiaries.

Section 6.5 Stock Option Exercise; Tax Withholding. Subject to such terms and conditions as shall be specified in an Award Agreement, a Stock Option may be exercised in whole or in part at any time during the term thereof by notice in the form required by the Company, together with payment of the aggregate exercise price therefor and applicable withholding tax. Payment of the exercise price shall be made in the manner set forth in an Award Agreement, unless otherwise provided by the Committee: (i) in cash or by cash equivalent acceptable to the Committee; (ii) by payment in shares of Common Stock that have been held by the Participant for at least six months (or such period as the Committee may deem appropriate for accounting purposes or otherwise), valued at the Fair Market Value of such shares on the date of exercise; (iii) through an open-market broker-assisted sales transaction pursuant to which the Company is promptly delivered the amount of proceeds necessary to satisfy the exercise price; (iv) by a

combination of the foregoing methods; or (v) by such other method as may be approved by the Committee and set forth in an Award Agreement. In addition to and at the time of payment of the exercise price, the Participant shall pay to the Company the full amount of any and all applicable income tax, employment tax and other amounts required to be withheld in connection with such exercise, payable under such of the methods described above for the payment of the exercise price as may be approved by the Committee and set forth in an Award Agreement.

Section 6.6 Limited Transferability of Nonqualified Stock Options. All Stock Options shall be nontransferable except (i) upon the Participant's death, in accordance with Section 14.3 hereof or (ii) in the case of Nonqualified Stock Options only, for the transfer of all or part of the Stock Option to a Participant's "family member" (as defined for purposes of the Form S-8 registration statement under the Securities Act of 1933), or as otherwise permitted by the Committee, in each case as may be approved by the Committee in its discretion at the time of proposed transfer. The transfer of a Nonqualified Stock Option may be subject to such terms and conditions as the Committee may in its discretion impose from time to time. Subsequent transfers of a Nonqualified Stock Option shall be prohibited other than in accordance with Section 14.3 hereof.

# Section 6.7 Additional Rules for Incentive Stock Options.

- (i) Eligibility. An Incentive Stock Option may only be granted to an Eligible Person who is considered an employee under Treasury Regulation §1.421-7(h) of the Company or any Subsidiary.
- (ii) Annual Limits. No Incentive Stock Option shall be granted to an Eligible Person as a result of which the aggregate Fair Market Value (determined as of the Date of Grant) of the stock with respect to which Incentive Stock Options are exercisable for the first time in any calendar year under the Plan and any other stock option plans of the Company or any Subsidiary would exceed \$100,000, determined in accordance with section 422(d) of the Code. This limitation shall be applied by taking Incentive Stock Options into account in the order in which granted.
- (iii) Ten Percent Stockholders. If a Stock Option granted under the Plan is intended to be an Incentive Stock Option, and if the Participant, at the time of grant, owns stock possessing ten percent or more of the total combined voting power of all classes of Common Stock of the Company or any Subsidiary, then (A) the Stock Option exercise price per share shall in no event be less than 110 percent of the Fair Market Value of the Common Stock on the date of such grant and (B) such Stock Option shall not be exercisable after the expiration of five years following the date such Stock Option is granted.
- (iv) Termination of Employment. An Award of an Incentive Stock Option may provide that such Stock Option may be exercised not later than 3 months following termination of employment of the Participant with the Company and all Subsidiaries, or not later than one year following death or a permanent and total disability within the meaning of section 22(e)(3) of the Code, as and to the extent determined by the Committee to comply with the requirements of section 422 of the Code.
- (v) Other Terms and Conditions; Nontransferability. Any Incentive Stock Option granted hereunder shall contain such additional terms and conditions, not inconsistent with the terms of the Plan, as are deemed necessary or desirable by the Committee, which terms, together with the terms of the Plan, shall be intended and interpreted to cause such Incentive Stock Option to qualify as an "incentive stock option" under section 422 of the Code. An Award Agreement for an Incentive Stock Option may provide that such Stock Option shall be treated as a Nonqualified Stock Option to the extent that certain requirements applicable to "incentive stock options" under the Code shall not be satisfied. An Incentive Stock Option shall by its terms be nontransferable other than by will or by the laws of descent and distribution, and shall be exercisable during the lifetime of a Participant only by such Participant.

(vi) Disqualifying Dispositions. If shares of Common Stock acquired by exercise of an Incentive Stock Option are disposed of within two years following the Date of Grant or one year following the transfer of such shares to the Participant upon exercise, the Participant shall, promptly following such disposition, notify the Company in writing of the date and terms of such disposition and provide such other information regarding the disposition as the Company may reasonably require.

**Section 6.8 Repricing of Stock Options Prohibited**. Subject to the anti-dilution adjustment provisions contained in Section 4.2 hereof, without the prior approval of the Company's stockholders, evidenced by a majority of votes cast, neither the Committee nor the Board shall cause the cancellation, substitution or amendment of a Stock Option that would have the effect of reducing the exercise price of such a Stock Option previously granted under the Plan, or otherwise approve any modification to such a Stock Option that would be treated as a "repricing."

#### 7. STOCK APPRECIATION RIGHTS

**Section 7.1 Grant of Stock Appreciation Rights**. A Stock Appreciation Right may be granted to any Eligible Person selected by the Committee. Stock Appreciation Rights may be granted on a basis that allows for the exercise of the right by the Participant or that provides for the automatic payment of the right upon a specified date or event. The maximum number of shares of Common Stock that may be subject to Stock Appreciation Rights granted to any Participant during any calendar year shall be limited to 750,000 shares (subject to adjustment as provided in Section 4.2 hereof).

Section 7.2 Freestanding Stock Appreciation Rights. A Stock Appreciation Right may be granted without any related Stock Option. The Committee shall in its discretion prescribe the time or times at which, or the conditions upon which, a Stock Appreciation Right or portion thereof shall become vested and/or exercisable. The requirements for vesting and exercisability of a Stock Appreciation Right may be based on the continued Service of a Participant with the Company or a Subsidiary for a specified time period (or periods) or on the attainment of a specified performance goal (or goals) established by the Committee in its discretion. A Stock Appreciation Right will be exercisable or payable at such time or times as determined by the Committee, provided that the maximum term of a Stock Appreciation Right shall be ten years from the Date of Grant. The Committee may, in its discretion, accelerate the vesting or exercisability of any Stock Appreciation Right at any time. The base price of a Stock Appreciation Right granted without any related Stock Option shall be determined by the Committee in its sole discretion; provided, however, that the base price per share of any such freestanding Stock Appreciation Right shall not be less than 100 percent of the Fair Market Value of the shares of Common Stock on the Date of Grant.

Section 7.3 Tandem Stock Option/Stock Appreciation Rights. A Stock Appreciation Right may be granted in tandem with a Stock Option, either at the time of grant or at any time thereafter during the term of the Stock Option. A tandem Stock Option/Stock Appreciation Right will entitle the holder to elect, as to all or any portion of the number of shares subject to the Award, to exercise either the Stock Option or the Stock Appreciation Right, resulting in the reduction of the corresponding number of shares subject to the right so exercised as well as the tandem right not so exercised. A Stock Appreciation Right granted in tandem with a Stock Option hereunder shall have a base price per share equal to the per share exercise price of the Stock Option, will be vested and exercisable at the same time or times that a related Stock Option is vested and exercisable, and will expire no later than the time at which the related Stock Option expires.

Section 7.4 Payment of Stock Appreciation Rights. A Stock Appreciation Right will entitle the holder, upon exercise or other payment of the Stock Appreciation Right, as applicable, to receive an amount determined by multiplying: (i) the excess of the Fair Market Value of a share of Common Stock on the date of exercise or payment of the Stock Appreciation Right over the base price of such Stock Appreciation Right, by (ii) the number of shares as to which such Stock Appreciation Right is exercised or paid. Payment of the amount determined under the foregoing may be made, as approved by the Committee and set forth in the Award Agreement, in shares of Common Stock valued at their Fair Market Value on the date of exercise or payment, in cash, or in a combination of shares of Common Stock and cash, subject to applicable tax withholding requirements.

**Section 7.5 Repricing of Stock Appreciation Rights Prohibited**. Subject to the anti-dilution adjustment provisions contained in Section 4.2 hereof, without the prior approval of the Company's stockholders, evidenced by a majority of votes cast, neither the Committee nor the Board shall cause the cancellation, substitution or amendment of a Stock Appreciation Right that would have the effect of reducing the base price of such a Stock Appreciation Right previously granted under the Plan, or otherwise approve any modification to such a Stock Appreciation Right that would be treated as a "repricing."

#### 8. RESTRICTED STOCK AWARDS

**Section 8.1 Grant of Restricted Stock Awards**. A Restricted Stock Award may be granted to any Eligible Person selected by the Committee. The Committee may require the payment by the Participant of a specified purchase price in connection with any Restricted Stock Award. The Committee may grant Restricted Stock Awards that are Section 162(m) Awards, as well as Restricted Stock Awards that are not Section 162(m) Awards. The maximum number of shares of Common Stock that may be subject to Restricted Stock Awards granted to a Participant during any one calendar year shall be limited to 250,000 shares (subject to adjustment as provided in Section 4.2 hereof).

Section 8.2 Vesting Requirements. The restrictions imposed on shares of Common Stock granted under a Restricted Stock Award shall lapse in accordance with the vesting requirements specified by the Committee in the Award Agreement. The requirements for vesting of a Restricted Stock Award may be based on the continued Service of the Participant with the Company or its Subsidiaries for a specified time period (or periods) or on the attainment of a specified performance goal (or goals) established by the Committee in its discretion. The Committee may, in its discretion, accelerate the vesting of a Restricted Stock Award at any time. In the case of any Restricted Stock Award that is a Section 162(m) Award, any such performance-based vesting requirements shall be based upon the performance criteria identified in Section 12.2 hereof, and the terms of the Award shall otherwise comply with the requirements described in Section 12.3 hereof. If the vesting requirements of a Restricted Stock Award shall not be satisfied, the Award shall be forfeited and the shares of Common Stock subject to the Award shall be returned to the Company. In the event that the Participant paid any purchase price with respect to such forfeited shares, unless otherwise provided by the Committee in an Award Agreement, the Company will refund to the Participant the lesser of (i) such purchase price and (ii) the Fair Market Value of such shares on the date of forfeiture.

Section 8.3 Restrictions. Shares granted under any Restricted Stock Award may not be transferred, assigned or subject to any encumbrance, pledge, or charge until all applicable restrictions are removed or have expired, unless otherwise allowed by the Committee. Failure to satisfy any applicable restrictions shall result in the shares subject to the Restricted Stock Award being forfeited and returned to the Company. In the event that the Participant paid any purchase price with respect to such forfeited shares, unless otherwise provided by the Committee in an Award Agreement, the Company will refund to the Participant the lesser of (i) such purchase price and (ii) the Fair Market Value of such shares on the date of forfeiture. The Committee may require in an Award Agreement that certificates representing the shares granted under a Restricted Stock Award bear a legend making appropriate reference to the restrictions imposed, and that certificates representing the shares granted or sold under a Restricted Stock Award will remain in the physical custody of an escrow holder until all restrictions are removed or have expired.

**Section 8.4 Rights as Stockholder**. Subject to the foregoing provisions of this Section 8 and the applicable Award Agreement, the Participant shall have all rights of a stockholder with respect to the shares granted to the Participant under a Restricted Stock Award, including the right to vote the shares and receive all dividends and other distributions paid or made with respect thereto, unless the Committee determines otherwise at the time the Restricted Stock Award is granted. The Committee may provide in an Award Agreement for the payment of dividends and distributions to the Participant at such times as paid to stockholders generally or at the times of vesting or other payment of the Restricted Stock Award.

**Section 8.5 Section 83(b) Election.** If a Participant makes an election pursuant to section 83(b) of the Code with respect to a Restricted Stock Award, the Participant shall file, within 30 days following the Date of Grant, a copy of such election with the Company and with the Internal Revenue Service, in accordance with the regulations under section 83 of the Code. The Committee may provide in an Award Agreement that the Restricted Stock Award is conditioned upon the Participant's making or refraining from making an election with respect to the Award under section 83(b) of the Code.

#### 9. STOCK UNIT AWARDS

Section 9.1 Grant of Stock Unit Awards. A Stock Unit Award may be granted to any Eligible Person selected by the Committee. The value of each stock unit under a Stock Unit Award is equal to the Fair Market Value of the Common Stock on the applicable date or time period of determination, as specified by the Committee. The Committee may grant Stock Unit Awards that are Section 162(m) Awards, as well as Stock Unit Awards that are not Section 162(m) Awards. The maximum number of units that may be subject to Stock Unit Awards granted to a Participant during any one calendar year shall be limited to 250,000 units (subject to adjustment as provided in Section 4.2 hereof). A Stock Unit Award shall be subject to such restrictions and conditions as the Committee shall determine. A Stock Unit Award may be granted together with a dividend equivalent right with respect to the shares of Common Stock subject to the Award, which may be accumulated and may be deemed reinvested in additional stock units, as determined by the Committee in its discretion.

Section 9.2 Vesting of Stock Unit Awards. On the Date of Grant, the Committee shall, in its discretion, determine any vesting requirements with respect to a Stock Unit Award, which shall be set forth in the Award Agreement. The requirements for vesting of a Stock Unit Award may be based on the continued Service of the Participant with the Company or its Subsidiaries for a specified time period (or periods) or on the attainment of a specified performance goal (or goals) established by the Committee in its discretion. The Committee may, in its discretion, accelerate the vesting of a Stock Unit Award at any time. In the case of any Stock Unit Award that is a Section 162(m) Award, any such performance-based vesting requirements shall be based upon the performance criteria identified in Section 12.2 hereof, and the terms of the Award shall otherwise comply with the requirements described in Section 12.3 hereof. A Stock Unit Award may also be granted on a fully vested basis, with a deferred payment date as may be determined by the Committee or elected by the Participant in accordance with the rules established by the Committee.

Section 9.3 Payment of Stock Unit Awards. A Stock Unit Award shall become payable to a Participant at the time or times determined by the Committee and set forth in the Award Agreement, which may be upon or following the vesting of the Award. Payment of a Stock Unit Award may be made, at the discretion of the Committee, in cash or in shares of Common Stock, or in a combination thereof, subject to applicable tax withholding requirements. Any cash payment of a Stock Unit Award shall be made based upon the Fair Market Value of the Common Stock, determined on such date or over such time period as determined by the Committee. In the case of a Participant who is a "specified employee" as defined in Section 409A of the Code at the time of any payment of a Stock Unit Award upon the Participant's termination of Service, the payments under the Stock Unit Award shall be deferred until the date that is six months following the Participant's termination of Service to the extent necessary to comply with Section 409A of the Code, with the terms of such deferral and payment to be made in the manner determined by the Committee and set forth in the Award Agreement.

**Section 9.4 No Rights as Stockholder**. The Participant shall not have any rights as a stockholder with respect to the shares subject to a Stock Unit Award until such time as shares of Common Stock are delivered to the Participant pursuant to the terms of the Award Agreement.

# 10. STOCK AWARDS

Section 10.1 Grant of Stock Awards. A Stock Award may be granted to any Eligible Person selected by the Committee. A Stock Award may be granted for past services, in lieu of bonus or other cash compensation, as directors' compensation or for any other valid purpose as determined by the Committee. A Stock Award granted to an Eligible Person represents shares of Common Stock that are issued free of restrictions on transfer and free of forfeiture conditions and to which such Eligible Person is entitled all incidents of ownership, except as otherwise provided in the Plan and the Award Agreement. The Committee may, in connection with any Stock Award, require the payment of a specified purchase price. The Committee may grant Stock Awards that are Section 162(m) Awards, as well as Stock Awards that are not Section 162(m) Awards. The maximum number of shares of Common Stock that may be subject to Stock Awards granted to a Participant during any one calendar year shall be limited to 250,000 shares (subject to adjustment as provided in Section 4.2 hereof).

**Section 10.2 Rights as Stockholder.** Subject to the foregoing provisions of this Section 10 and the applicable Award Agreement, upon the issuance of the shares of Common Stock under a Stock Award, the Participant shall have all rights of a stockholder with respect to the shares of Common Stock, including the right to vote the shares and receive all dividends and other distributions paid or made with respect thereto.

#### 11. PERFORMANCE AWARDS

Section 11.1 Grant of Performance Awards. The Committee may grant Performance Awards under the Plan, which shall represent the right to receive a payment in cash if performance goals established by the Committee for a performance period are satisfied. The Committee may grant Performance Awards that are Section 162(m) Awards, as well as Performance Awards that are not Section 162(m) Awards. At the time a Performance Award is granted, the Committee shall determine, in its sole discretion, the applicable performance period and performance goals to be achieved during the performance period, as well as such other conditions as the Committee deems appropriate. The Committee may also determine a target payment amount or a range of payment amounts for each Award. The performance goals applicable to a Performance Award grant may be subject to adjustments as the Committee shall deem appropriate to reflect significant unforeseen events, such as changes in law, accounting practices or unusual or nonrecurring items or occurrences. The Committee's authority to make such adjustments shall be subject to such limitations as the Committee deems appropriate in the case of a Performance Award that is a Section 162(m) Award. In the case of any Performance Award that is a Section 162(m) Award, performance goals shall be based upon the performance criteria identified in Section 12.2 hereof, and the terms of the Award shall otherwise comply with the requirements described in Section 12.3 hereof. The maximum amount of cash compensation that may be paid to a Participant during any one calendar year under Performance Awards shall be \$1 million.

Section 11.2 Payment of Performance Awards. At the end of the performance period, the Committee shall determine the extent to which performance goals have been attained, or a degree of achievement between minimum and maximum levels, in order to establish the level of payment to be made, if any. Payments of Performance Awards shall generally be made as soon as practicable following the end of the performance period, subject to any tax withholding requirements. In the case of a Participant who is a "specified employee" as defined in Section 409A of the Code at the time of any payment of a Performance Award upon the Participant's termination of Service, the payments under the Performance Award shall be deferred until the date that is six months following the Participant's termination of Service to the extent necessary to comply with Section 409A of the Code, with the terms of such deferral and payment to be made in the manner determined by the Committee and set forth in the Award Agreement.

#### 12. SECTION 162(M) AWARDS

**Section 12.1 Awards**. Awards of Stock Options and Stock Appreciation Rights granted under the Plan are intended by their terms to qualify as Section 162(m) Awards. Restricted Stock Awards, Stock Unit Awards, Stock Awards and Performance Awards granted under the Plan may qualify as Section 162(m) Awards if the Awards are granted or become payable or vested based upon pre-established performance goals in accordance with this Section 12.

Section 12.2 Performance Criteria. In the case of a Restricted Stock Award, Stock Unit Award, Stock Award or Performance Award that is intended to be a Section 162(m) Award, the performance criteria upon which the grant, payment or vesting may be based shall be limited to one or more of the following performance measures, which may be applied with respect to the Company, any Subsidiary or any business unit: (i) total stockholder return; (ii) stock price increase; (iii) return on equity; (iv) return on capital; (v) return on investment; (vi) earnings per share, diluted or basic; (vii) EBIT (earnings before interest and taxes); (viii) EBITDA (earnings before interest, taxes, depreciation and amortization); (ix) cash flow (including operating cash flow, free cash flow, discounted cash flow, and cash flow in excess of costs of capital); (x) net or gross revenue; (xi) operating expenses; (xii) gross or operating margin; (xiii) execution of a corporate collaboration agreement relating to a product candidate of the Company; (xiv) acceptance by the U.S. Food and Drug Administration ("FDA") or a comparable foreign regulatory authority of a final New Drug Application, a Biologic License Application or similar document; (xv) approval for marketing of a product candidate of the Company by the FDA or a comparable foreign regulatory authority; (xvi) obtaining a specified level of financing for the Company, as determined by the Committee, including through government grants (or similar awards) and the issuance of securities; and (xvii) commencement of a particular stage of clinical trials for a product candidate of the Company. The foregoing performance criteria shall have any reasonable definitions that the Committee may specify, which may include or exclude any items specified by the Committee, including but not limited to any or all of the following items: discontinued operations, extraordinary, unusual or non-recurring items, effects of accounting changes, effects of currency or interest rate fluctuations, effects of financing activities (e.g., effect on earnings per share of issuing convertible debt securities), changes in tax rates, expenses for restructuring or productivity initiatives, litigation losses, non-operating items, effects of acquisitions or divestitures and changes of law or regulation affecting the Company's business. The foregoing performance measures may be determined on an absolute basis or relative to internal goals or relative to levels attained in prior years, or related to other companies or indices, or as ratios expressing relationships between two or more performance measures. In the case of Awards that are not Section 162(m) Awards, the Committee may designate performance criteria from among the foregoing or such other performance criteria as it shall determine in its sole discretion.

Section 12.3 Section 162(m) Requirements. In the case of a Restricted Stock Award, Stock Unit Award, Stock Award or Performance Award that is intended to be a Section 162(m) Award, the Committee shall make such determinations with respect to an Award as required by section 162(m) of the Code within 90 days after the beginning of the performance period (or such other time period as is required under section 162(m) of the Code). As and to the extent required by section 162(m) of the Code, the terms of an Award that is a Section 162(m) Award must state, in terms of an objective formula or standard, the method of computing the amount of compensation payable under the Award, and must preclude discretion to increase the amount of compensation payable under the terms of the Award (but may allow the Committee discretion to decrease the amount of compensation payable).

#### 13. CHANGE IN CONTROL

Section 13.1 Effect of Change in Control. The Committee may, at the time of the grant of an Award and as set forth in an Award Agreement, provide for the effect of a "Change in Control" on an Award. Such provisions may include any one or more of the following: (i) the acceleration or extension of time periods for purposes of exercising, vesting in, or realizing gain from any Award, (ii) the elimination or modification of performance or other conditions related to the payment or other rights under an Award, (iii) provision for the cash settlement of an Award for an equivalent cash value, as determined by the Committee, or (iv) such other modification or adjustment to an Award as the Committee deems appropriate to maintain and protect the rights and interests of Participants upon or following a Change in Control. To the extent necessary for compliance with Section 409A of the Code, an Award Agreement shall provide that an Award subject to the requirements of Section 409A that would otherwise become payable upon a Change in Control shall only become payable to the extent that the requirements for a "change in control" for purposes of Section 409A have been satisfied.

**Section 13.2 Definition of Change in Control.** For purposes of the Plan, unless otherwise defined in an Award Agreement, a "Change in Control" shall be deemed to have occurred upon:

- (i) a change in the composition of the Board such that during any period of two consecutive years, individuals who at the beginning of such period constitute the Board, and any new director (other than a director designated by a person who has entered into an agreement with the Company to effect a transaction described in clause (ii) or (iii) of this Section 13.2) whose election by the Board or nomination for election by the Company's stockholders was approved by a vote of at least two-thirds of the directors then still in office who either were directors at the beginning of the period or whose election or nomination for election was previously so approved, cease for any reason to constitute at least a majority of the members thereof;
- (ii) the consummation of a merger, consolidation, reorganization or similar corporate transaction, whether or not the Company is the surviving corporation in such transaction, in which outstanding shares of Common Stock are converted into (A) shares of stock of another company, other than a conversion into shares of voting common stock of the successor corporation (or a holding company thereof) representing more than 50% of the voting power of all capital stock thereof outstanding immediately after the merger or consolidation, or (B) other securities (of either the Company or another company) or cash or other property;
- (iii) any "Person" (as such term is used in Sections 13(d)(3) and 14(d)(2) of the Exchange Act), except that such term shall not include (A) the Company, (B) a trustee or other fiduciary holding securities under an employee benefit plan of the Company, (C) an underwriter temporarily holding securities pursuant to an offering of such securities, or (D) a corporation owned, directly or indirectly, by the stockholders of the Company in substantially the same proportions as their ownership of stock of the Company, who is or becomes the "Beneficial Owner" (as defined in Rule 13d-3 under the Exchange Act), directly or indirectly, of securities of the Company (not including in the securities Beneficially Owned by such Person any securities acquired directly from the Company) representing 30% or more of the voting power of all capital stock thereof outstanding, excluding any Person who is an officer or director of the Company or who becomes such a Beneficial Owner in connection with a transaction described in clause (ii) of this Section 13.2; or
- (iv) the consummation of (A) the sale or other disposition of all or substantially all of the assets of the Company, or (B) a complete liquidation or dissolution of the Company.

#### 14. GENERAL PROVISIONS

Section 14.1 Award Agreement. To the extent deemed necessary by the Committee, an Award under the Plan shall be evidenced by an Award Agreement in a written or electronic form approved by the Committee setting forth the number of shares of Common Stock or units subject to the Award, the exercise price, base price, or purchase price of the Award, the time or times at which an Award will become vested, exercisable or payable and the term of the Award. The Award Agreement may also set forth the effect on an Award of termination of Service under certain circumstances. The Award Agreement shall be subject to and incorporate, by reference or otherwise, all of the applicable terms and conditions of the Plan, and may also set forth other terms and conditions applicable to the Award as determined by the Committee consistent with the limitations of the Plan. Award Agreements evidencing Incentive Stock Options shall contain such terms and conditions as may be necessary to meet the applicable provisions of section 422 of the Code. The grant of an Award under the Plan shall not confer any rights upon the Participant holding such Award other than such terms, and subject to such conditions, as are specified in the Plan as being applicable to such type of Award (or to all Awards) or as are expressly set forth in the Award Agreement. The Committee need not require the execution of an Award Agreement by a Participant, in which case, acceptance of the Award by the Participant shall constitute agreement by the Participant to the terms, conditions, restrictions and limitations set forth in the Plan and the Award Agreement as well as the administrative guidelines of the Company in effect from time to time.

Section 14.2 Forfeiture Events/Representations. The Committee may specify in an Award Agreement at the time of the Award that the Participant's rights, payments and benefits with respect to an Award shall be subject to reduction, cancellation, forfeiture or recoupment upon the occurrence of certain specified events, in addition to any otherwise applicable vesting or performance conditions of an Award. Such events shall include, but shall not be limited to, termination of Service for cause, violation of material Company policies, breach of noncompetition, confidentiality or other restrictive covenants that may apply to the Participant, or other conduct by the Participant that is detrimental to the business or reputation of the Company. The Committee may also specify in an Award Agreement that the Participant's rights, payments and benefits with respect to an Award shall be conditioned upon the Participant making a representation regarding compliance with noncompetition, confidentiality or other restrictive covenants that may apply to the Participant and providing that the Participant's rights, payments and benefits with respect to an Award shall be subject to reduction, cancellation, forfeiture or recoupment on account of a breach of such representation.

Section 14.3 No Assignment or Transfer; Beneficiaries. Except as provided in Section 6.6 hereof, Awards under the Plan shall not be assignable or transferable by the Participant, except by will or by the laws of descent and distribution, and shall not be subject in any manner to assignment, alienation, pledge, encumbrance or charge. Notwithstanding the foregoing, the Committee may provide in an Award Agreement that the Participant shall have the right to designate a beneficiary or beneficiaries who shall be entitled to any rights, payments or other benefits specified under an Award following the Participant's death. During the lifetime of a Participant, an Award shall be exercised only by such Participant or such Participant's guardian or legal representative. In the event of a Participant's death, an Award may, to the extent permitted by the Award Agreement, be exercised by the Participant's beneficiary as designated by the Participant in the manner prescribed by the Committee or, in the absence of an authorized beneficiary designation, by the legatee of such Award under the Participant's will or by the Participant's estate in accordance with the Participant's will or the laws of descent and distribution, in each case in the same manner and to the same extent that such Award was exercisable by the Participant on the date of the Participant's death.

**Section 14.4 Deferrals of Payment.** The Committee may in its discretion permit a Participant to defer the receipt of payment of cash or delivery of shares of Common Stock that would otherwise be due to the Participant by virtue of the exercise of a right or the satisfaction of vesting or other conditions with respect to an Award. If any such deferral is to be permitted by the Committee, the Committee shall establish rules and procedures relating to such deferral in a manner intended to comply with the requirements of Section 409A of the Code, including, without limitation, the time when an election to defer may be made, the time period of the deferral and the events that would result in payment of the deferred amount, the interest or other earnings attributable to the deferral and the method of funding, if any, attributable to the deferred amount.

**Section 14.5 Rights as Stockholder**. A Participant shall have no rights as a holder of shares of Common Stock with respect to any unissued securities covered by an Award until the date the Participant becomes the holder of record of such securities. Except as provided in Section 4.2 hereof, no adjustment or other provision shall be made for dividends or other stockholder rights, except to the extent that the Award Agreement provides for dividend payments or dividend equivalent rights.

Section 14.6 Employment or Service. Nothing in the Plan, in the grant of any Award or in any Award Agreement shall confer upon any Eligible Person or Participant any right to continue in the Service of the Company or any of its Subsidiaries, or interfere in any way with the right of the Company or any of its Subsidiaries to terminate the employment or other service relationship of an Eligible Person or Participant for any reason at any time.

Section 14.7 Securities Laws. No shares of Common Stock will be issued or transferred pursuant to an Award unless and until all then applicable requirements imposed by Federal and state securities and other laws, rules and regulations and by any regulatory agencies having jurisdiction, and by any exchanges upon which the shares of Common Stock may be listed, have been fully met. As a condition precedent to the issuance of shares pursuant to the grant or exercise of an Award, the Company may require the Participant to take any reasonable action to meet such requirements. The Committee may impose such conditions on any shares of Common Stock issuable under the Plan as it may deem advisable, including, without limitation, restrictions under the Securities Act of 1933, as amended, under the requirements of any exchange upon which such shares of the same class are then listed, and under any blue sky or other securities laws applicable to such shares. The Committee may also require the Participant to represent and warrant at the time of issuance or transfer that the shares of Common Stock are being acquired only for investment purposes and without any current intention to sell or distribute such shares.

**Section 14.8 Tax Withholding.** The Participant shall be responsible for payment of any taxes or similar charges required by law to be withheld from an Award or an amount paid in satisfaction of an Award, which shall be paid by the Participant on or prior to the payment or other event that results in taxable income in respect of an Award. The Award Agreement may specify the manner in which the withholding obligation shall be satisfied with respect to the particular type of Award.

**Section 14.9 Unfunded Plan.** The adoption of the Plan and any reservation of shares of Common Stock or cash amounts by the Company to discharge its obligations hereunder shall not be deemed to create a trust or other funded arrangement. Except upon the issuance of Common Stock pursuant to an Award, any rights of a Participant under the Plan shall be those of a general unsecured creditor of the Company, and neither a Participant nor the Participant's permitted transferees or estate shall have any other interest in any assets of the Company by virtue of the Plan. Notwithstanding the foregoing, the Company shall have the right to implement or set aside funds in a grantor trust, subject to the claims of the Company's creditors or otherwise, to discharge its obligations under the Plan.

Section 14.10 Other Compensation and Benefit Plans. The adoption of the Plan shall not affect any other stock incentive or other compensation plans in effect for the Company or any Subsidiary, nor shall the Plan preclude the Company from establishing any other forms of stock incentive or other compensation or benefit program for employees of the Company or any Subsidiary. The amount of any compensation deemed to be received by a Participant pursuant to an Award shall not constitute includable compensation for purposes of determining the amount of benefits to which a Participant is entitled under any other compensation or benefit plan or program of the Company or any Subsidiary, including, without limitation, under any bonus, pension, profit-sharing, life insurance, salary continuation or severance benefits plan, except to the extent specifically provided by the terms of any such plan.

**Section 14.11 Plan Binding on Transferees**. The Plan shall be binding upon the Company, its transferees and assigns, and the Participant, the Participant's executor, administrator and permitted transferees and beneficiaries.

**Section 14.12 Severability**. If any provision of the Plan or any Award Agreement shall be determined to be illegal or unenforceable by any court of law in any jurisdiction, the remaining provisions hereof and thereof shall be severable and enforceable in accordance with their terms, and all provisions shall remain enforceable in any other jurisdiction.

Section 14.13 Foreign Jurisdictions. The Committee may adopt, amend and terminate such arrangements and grant such Awards, not inconsistent with the intent of the Plan, as it may deem necessary or desirable to comply with any tax, securities, regulatory or other laws of other jurisdictions with respect to Awards that may be subject to such laws. The terms and conditions of such Awards may vary from the terms and conditions that would otherwise be required by the Plan solely to the extent the Committee deems necessary for such purpose. Moreover, the Board may approve such supplements to or amendments, restatements or alternative versions of the Plan, not inconsistent with the intent of the Plan, as it may consider necessary or appropriate for such purposes, without thereby affecting the terms of the Plan as in effect for any other purpose.

Section 14.14 Substitute Awards in Corporate Transactions. Nothing contained in the Plan shall be construed to limit the right of the Committee to grant Awards under the Plan in connection with the acquisition, whether by purchase, merger, consolidation or other corporate transaction, of the business or assets of any corporation or other entity. Without limiting the foregoing, the Committee may grant Awards under the Plan to an employee or director of another corporation who becomes an Eligible Person by reason of any such corporate transaction in substitution for awards previously granted by such corporation or entity to such person. The terms and conditions of the substitute Awards may vary from the terms and conditions that would otherwise be required by the Plan solely to the extent the Committee deems necessary for such purpose. Any shares of Common Stock subject to these substitute Awards shall not be counted against any of the maximum share limitations set forth in the Plan.

**Section 14.15 Governing Law**. The Plan and all rights hereunder shall be subject to and interpreted in accordance with the laws of the State of Delaware, without reference to the principles of conflicts of laws, and to applicable Federal securities laws.

Section 14.16 Section 409A Compliance. To the extent applicable, it is intended that the Plan and all Awards hereunder comply with the requirements of Section 409A of the Code, and the Plan and all Award Agreements shall be interpreted and applied by the Committee in a manner consistent with this intent in order to avoid the imposition of any additional tax under Section 409A of the Code. In the event that any provision of the Plan or an Award Agreement is determined by the Committee to not comply with the applicable requirements of Section 409A of the Code, the Committee shall have the authority to take such actions and to make such interpretations or changes to the Plan or an Award Agreement as the Committee deems necessary to comply with such requirements, provided that the Committee shall act in a manner that is intended to preserve the economic value of the Award to the Participant. Notwithstanding the foregoing or anything elsewhere in the Plan or an Award Agreement to the contrary, if a Participant is a "specified employee" as defined in Section 409A of the Code at the time of termination of Service with respect to an Award, then solely to the extent necessary to avoid the imposition of any additional tax under Section 409A of the Code, the commencement of any payments or benefits under the Award shall be deferred until the date that is six months following the Participant's termination of Service (or such other period as required to comply with Section 409A).

#### 15. EFFECTIVE DATE; AMENDMENT AND TERMINATION

Section 15.1 Effective Date. The Plan became effective following its adoption by the Progenics Board and its approval by the Progenics' stockholders on the date of the 2005 Annual Meeting of Stockholders. The term of the Plan shall be ten (10) years from the date of such adoption by the Progenics Board, subject to Section 15.3 hereof. The Plan was amended and restated following its adoption by the Progenics Board and its approval by the Progenics' stockholders on the date of the 2007 Annual Meeting of Stockholders, and was amended by the Progenics Board to extend its term as noted in Section 15.3 subject to stockholder approval, which occurred at the 2014 Annual Meeting of Stockholders. The Plan as amended in connection with the Merger became effective following its adoption by the Board and the closing of the Merger on June 19, 2020.

Section 15.2 Amendment. The Board may at any time and from time to time and in any respect, amend or modify the Plan. The Board may seek the approval of any amendment or modification by the Company's stockholders to the extent it deems necessary or advisable in its discretion for purposes of compliance with section 162(m) or section 422 of the Code, the listing requirements of the NASDAQ or other exchange or securities market or for any other purpose. No amendment or modification of the Plan shall adversely affect any Award theretofore granted without the consent of the Participant or the permitted transferee of the Award. Notwithstanding the foregoing and notwithstanding anything to the contrary in the Plan, the Board may amend the Plan and any outstanding Award Agreement solely to comply with any new regulations or other guidance from the Internal Revenue Service under section 409A of the Code without the consent of the Participant or the permitted transferee of the Award.

**Section 15.3 Termination**. The Plan shall terminate on March 25, 2024, which is the date immediately preceding the 10th anniversary of the date of the approval by the Progenics Board of an amendment to extend the term of the Plan to such date. The Board may, in its discretion and at any earlier date, terminate the Plan. Notwithstanding the foregoing, no termination of the Plan shall adversely affect any Award theretofore granted without the consent of the Participant or the permitted transferee of the Award.

# LANTHEUS HOLDINGS, INC. 2018 PERFORMANCE INCENTIVE PLAN (F/K/A PROGENICSPHARMACEUTICALS, INC. 2018 PERFORMANCE INCENTIVE PLAN)

(As amended, effective June 19, 2020)

#### 1. PURPOSE OF PLAN

The purpose of this Lantheus Holdings, Inc. 2018 Performance Incentive Plan (formerly known as Progenics Pharmaceuticals, Inc. 2018 Performance Incentive Plan) (this "**Plan**") of Lantheus Holdings, Inc., a Delaware corporation (the "**Corporation**"), is to promote the success of the Corporation by providing an additional means through the grant of awards to attract, motivate, retain and reward selected employees and other eligible persons and to enhance the alignment of the interests of the selected participants with the interests of the Corporation's stockholders.

#### 2. ELIGIBILITY

The Administrator (as such term is defined in Section 3.1) may grant awards under this Plan only to those persons that the Administrator determines to be Eligible Persons. An "Eligible Person" is any person who is either: (a) an officer (whether or not a director) or employee of the Corporation or one of its Subsidiaries; (b) a director of the Corporation or one of its Subsidiaries; or (c) an individual consultant or advisor who renders or has rendered bona fide services (other than services in connection with the offering or sale of securities of the Corporation or one of its Subsidiaries in a capital-raising transaction or as a market maker or promoter of securities of the Corporation or one of its Subsidiaries) to the Corporation or one of its Subsidiaries and who is selected to participate in this Plan by the Administrator; provided, however, that a person who is otherwise an Eligible Person under clause (c) above may participate in this Plan only if such participation would not adversely affect either the Corporation's eligibility to use Form S-8 to register under the Securities Act of 1933, as amended (the "Securities Act"), the offering and sale of shares issuable under this Plan by the Corporation or the Corporation's compliance with any other applicable laws; provided, further, that following the Merger, only the following persons will be eligible to receive awards under the Plan: (i) Eligible Persons that were employed by Progenics or its subsidiaries prior to the Merger; and (ii) Eligible Persons hired by the Corporation or one of its Subsidiaries following the Merger. An Eligible Person who has been granted an award (a "participant") may, if otherwise eligible, be granted additional awards if the Administrator shall so determine. As used herein, "Progenics" means Progenics Pharmaceuticals, Inc.; "Subsidiary" means any corporation or other entity a majority of whose outstanding voting stock or voting power is beneficially owned directly or indirectly by the Corporation; "Board" means the Board of Directors of the Corporation; "Merger" means the acquisition of all outstanding shares of common stock, par value \$0.0013 per share, of Progenics by the Corporation by way of a merger of Plato Merger Sub, Inc., a Delaware corporation and a wholly-owned subsidiary of the Corporation, with and into Progenics, with Progenics being the surviving corporation.

### 3. PLAN ADMINISTRATION

3.1 The Administrator. This Plan shall be administered by and all awards under this Plan shall be authorized by the Administrator. The "Administrator" means the Board or one or more committees (or subcommittees, as the case may be) appointed by the Board or another committee (within its delegated authority) to administer all or certain aspects of this Plan. Any such committee shall be comprised solely of one or more directors or such number of directors as may be required under applicable law. A committee may delegate some or all of its authority to another committee so constituted. The Board or a committee comprised solely of directors may also delegate, to the extent

permitted by Section 157(c) of the Delaware General Corporation Law and any other applicable law, to one or more officers of the Corporation, its authority under this Plan. The Board or another committee (within its delegated authority) may delegate different levels of authority to different committees or persons with administrative and grant authority under this Plan. Unless otherwise provided in the Bylaws of the Corporation or the applicable charter of any Administrator: (a) a majority of the members of the acting Administrator shall constitute a quorum, and (b) the vote of a majority of the members present assuming the presence of a quorum or the unanimous written consent of the members of the Administrator shall constitute action by the acting Administrator.

- **3.2 Powers of the Administrator.** Subject to the express provisions of this Plan, the Administrator is authorized and empowered to do all things necessary or desirable in connection with the authorization of awards and the administration of this Plan (in the case of a committee or delegation to one or more officers, within any express limits on the authority delegated to that committee or person(s)), including, without limitation, the authority to:
  - (a) determine eligibility and, from among those persons determined to be eligible, determine the particular Eligible Persons who will receive an award under this Plan;
  - (b) grant awards to Eligible Persons, determine the price (if any) at which securities will be offered or awarded and the number of securities to be offered or awarded to any of such persons (in the case of securities-based awards), determine the other specific terms and conditions of awards consistent with the express limits of this Plan, establish the installment(s) (if any) in which such awards shall become exercisable or shall vest (which may include, without limitation, performance and/or time-based schedules), or determine that no delayed exercisability or vesting is required, establish any applicable performance-based exercisability or vesting requirements, determine the circumstances in which any performance-based goals (or the applicable measure of performance) will be adjusted and the nature and impact of any such adjustment, determine the extent (if any) to which any applicable exercise and vesting requirements have been satisfied, establish the events (if any) on which exercisability or vesting may accelerate (which may include, without limitation, retirement and other specified terminations of employment or services, or other circumstances), and establish the events (if any) of termination, expiration or reversion of such awards;
  - (c) approve the forms of any award agreements (which need not be identical either as to type of award or among participants);
  - (d) construe and interpret this Plan and any agreements defining the rights and obligations of the Corporation, its Subsidiaries, and participants under this Plan, make any and all determinations under this Plan and any such agreements, further define the terms used in this Plan, and prescribe, amend and rescind rules and regulations relating to the administration of this Plan or the awards granted under this Plan;
  - (e) cancel, modify, or waive the Corporation's rights with respect to, or modify, discontinue, suspend, or terminate any or all outstanding awards, subject to any required consent under Section 8.6.5;

- (f) accelerate, waive or extend the vesting or exercisability, or modify or extend the term of, any or all such outstanding awards (in the case of options or stock appreciation rights, within the maximum ten-year term of such awards) in such circumstances as the Administrator may deem appropriate (including, without limitation, in connection with a retirement or other termination of employment or services, or other circumstances) subject to any required consent under Section 8.6.5;
- (g) adjust the number of shares of Common Stock subject to any award, adjust the price of any or all outstanding awards or otherwise waive or change previously imposed terms and conditions, in such circumstances as the Administrator may deem appropriate, in each case subject to Sections 4 and 8.6 (and subject to the no repricing provision below);
- (h) determine the date of grant of an award, which may be a designated date after but not before the date of the Administrator's action to approve the award (unless otherwise designated by the Administrator, the date of grant of an award shall be the date upon which the Administrator took the action approving the award);
- (i) determine whether, and the extent to which, adjustments are required pursuant to Section 7.1 hereof and take any other actions contemplated by Section 7 in connection with the occurrence of an event of the type described in Section 7;
- (j) acquire or settle (subject to Sections 7 and 8.6) rights under awards in cash, stock of equivalent value, or other consideration (subject to the no repricing provision below); and
  - (k) determine the fair market value of the Common Stock or awards under this Plan from time to time and/or the manner in which such value will be determined.
- **3.3 Prohibition on Repricing.** Notwithstanding anything to the contrary in Section 3.2 and except for an adjustment pursuant to Section 7.1 or a repricing approved by stockholders, in no case may the Administrator (1) amend an outstanding stock option or SAR to reduce the exercise price or base price of the award, (2) cancel, exchange, or surrender an outstanding stock option or SAR in exchange for an option or SAR with an exercise or base price that is less than the exercise or base price of the original award.
- 3.4 Binding Determinations. Any determination or other action taken by, or inaction of, the Corporation, any Subsidiary, or the Administrator relating or pursuant to this Plan (or any award made under this Plan) and within its authority hereunder or under applicable law shall be within the absolute discretion of that entity or body and shall be conclusive and binding upon all persons. Neither the Board nor any other Administrator, nor any member thereof or person acting at the direction thereof, shall be liable for any act, omission, interpretation, construction or determination made in good faith in connection with this Plan (or any award made under this Plan), and all such persons shall be entitled to indemnification and reimbursement by the Corporation in respect of any claim, loss, damage or expense (including, without limitation, attorneys' fees) arising or resulting therefrom to the fullest extent permitted by law and/or under any directors and officers liability insurance coverage that may be in effect from time to time. Neither the Board nor any other Administrator, nor any member thereof or person acting at the direction thereof, nor the Corporation or any of its Subsidiaries, shall be liable for any damages of a

participant should an option intended as an ISO (as defined below) fail to meet the requirements of the Internal Revenue Code of 1986, as amended (the "Code"), applicable to ISOs, should any other award(s) fail to qualify for any intended tax treatment, should any award grant or other action with respect thereto not satisfy Rule 16b-3 promulgated under the Securities Exchange Act of 1934, as amended, or otherwise for any tax or other liability imposed on a participant with respect to an award.

- **3.5 Reliance on Experts.** In making any determination or in taking or not taking any action under this Plan, the Administrator may obtain and may rely upon the advice of experts, including employees and professional advisors to the Corporation. No director, officer or agent of the Corporation or any of its Subsidiaries shall be liable for any such action or determination taken or made or omitted in good faith.
- **3.6 Delegation**. The Administrator may delegate ministerial, non-discretionary functions to individuals who are officers or employees of the Corporation or any of its Subsidiaries or to third parties.

# 4. SHARES OF COMMON STOCK SUBJECT TO THE PLAN; SHARE LIMITS

- **4.1 Shares Available.** Subject to the provisions of Section 7.1, the capital stock that may be delivered under this Plan shall be shares of the Corporation's authorized but unissued Common Stock and any shares of its Common Stock held as treasury shares. For purposes of this Plan, "Common Stock" shall mean the common stock of the Corporation and such other securities or property as may become the subject of awards under this Plan, or may become subject to such awards, pursuant to an adjustment made under Section 7.1.
- **4.2 Aggregate Share Limit.** The maximum number of shares of Common Stock that may be delivered pursuant to awards granted to Eligible Persons under this Plan (the "Share Limit") is equal to the sum of the following:
  - (1) 1,488,000 shares of Common Stock (this being the adjusted number upon assumption of the Plan in connection with the Merger)), plus
  - (2) the number of shares of Common Stock available for additional award grant purposes under the Corporation's 2005 Stock Incentive Plan (as amended effective June 17, 2014) (the "2005 Plan") as of the date of stockholder approval of this Plan (the "Stockholder Approval Date") (as adjusted to reflect the Merger) and determined immediately prior to the termination of the authority to grant new awards under the 2005 Plan as of the Stockholder Approval Date, plus
  - (3) the number of any shares subject to stock options granted under the 2005 Plan and outstanding on the Stockholder Approval Date which expire, or for any reason are cancelled or terminated, after the Stockholder Approval Date without being exercised (as adjusted to reflect the Merger) (which, for purposes of clarity, shall become available for award grants under this Plan on a one-for-one basis), plus
  - (4) the number of any shares subject to restricted stock and restricted stock unit awards granted under the 2005 Plan that are outstanding and unvested on the Stockholder Approval Date that are forfeited, terminated, cancelled or otherwise reacquired by the Corporation without having become vested (as adjusted to reflect the Merger), provided that in order to take the Full-Value Award ratio below into account, each share subject to any such award shall be credited as one and forty-two one-hundredths (1.42) shares when determining the number of shares that shall become available for new awards under this Plan.

provided that in no event shall the Share Limit exceed 4,371,992 shares, which is the sum of the 1,488,000 shares set forth above (this being the adjusted number upon assumption of the Plan in connection with the Merger), plus the number of shares available under the 2005 Plan for additional award grant purposes as of the Effective Date (as such term is defined in Section 8.6.1) (as adjusted to reflect the Merger), plus the aggregate number of shares subject to awards previously granted and outstanding under the 2005 Plan as of the Effective Date (as adjusted to reflect the Merger), with any shares subject to restricted stock and restricted stock unit awards outstanding under the 2005 Plan being taken into account based on the share-counting ratio for such awards under clause (4) above).

Shares issued in respect of any "Full-Value Award" granted under this Plan shall be counted against the foregoing Share Limit as one and forty-two one-hundredths (1.42) shares for every one share issued in connection with such award. (For example, if a stock bonus of 100 shares of Common Stock is granted under this Plan, 142 shares shall be counted against the Share Limit in connection with that award.) For this purpose, a "Full-Value Award" means any award that is <u>not</u> a stock option grant or a stock appreciation right grant.

- **4.3 Additional Share Limits.** The following limits also apply with respect to awards granted under this Plan. These limits are in addition to, not in lieu of, the aggregate Share Limit in Section 4.2.
  - (a) The maximum number of shares of Common Stock that may be delivered pursuant to options qualified as incentive stock options granted under this Plan is 1,488,000 shares (this being the adjusted number upon assumption of the Plan in connection with the Merger).
  - (b) Awards that are granted under this Plan during any one calendar year to any person who, on the grant date of the award, is a non-employee director are subject to the limits of this Section 4.3(b). The maximum number of shares of Common Stock subject to those awards that are granted under this Plan during any one calendar year to an individual who, on the grant date of the award, is a non-employee director is 31,000 shares (this being the adjusted number upon assumption of the Plan in connection with the Merger ); provided that this limit is 62,000 shares (this being the adjusted number upon assumption of the Plan in connection with the Merger ) as to (1) a non-employee director who is serving as the independent Chair of the Board or as a lead independent director at the time the applicable grant is made or (2) any new non-employee director for the calendar year in which the non-employee director is first elected or appointed to the Board. For purposes of this Section 4.3(b), a "non-employee director" is an individual who, on the grant date of the award, is a member of the Board who is not then an officer or employee of the Corporation or one of its Subsidiaries. The limits of this Section 4.3(b) do not apply to, and shall be determined without taking into account, any award granted to an individual who, on the grant date of the award, is an officer or employee of the Corporation or one of its Subsidiaries. The limits of this Section 4.3 (b) apply on an individual basis and not on an aggregate basis to all non-employee directors as a group.

- **4.4 Share-Limit Counting Rules.** The Share Limit shall be subject to the following provisions of this Section 4.4:
  - (a) Shares that are subject to or underlie awards granted under this Plan which expire or for any reason are cancelled or terminated, are forfeited, fail to vest, or for any other reason are not paid or delivered under this Plan shall not be counted against the Share Limit and shall be available for subsequent awards under this Plan.
  - (b) To the extent that shares of Common Stock are delivered pursuant to the exercise of a stock appreciation right granted under this Plan, the number of shares as to which such exercise relates shall be counted against the Share Limit. (For purposes of clarity, if a stock appreciation right relates to 100,000 shares and is exercised in full at a time when the payment due to the participant is 15,000 shares, 100,000 shares shall be counted against the Share Limit with respect to such exercise.)
  - (c) Shares that are exchanged by a participant or withheld by the Corporation as full or partial payment in connection with any award granted under this Plan, as well as any shares exchanged by a participant or withheld by the Corporation or one of its Subsidiaries to satisfy the tax withholding obligations related to any award granted under this Plan, shall be counted against the Share Limit and shall not be available for subsequent awards under this Plan.
  - (d) To the extent that an award granted under this Plan is settled in cash or a form other than shares of Common Stock, the shares that would have been delivered had there been no such cash or other settlement shall not be counted against the Share Limit and shall be available for subsequent awards under this Plan.
  - (e) The Corporation may not increase the Share Limit by repurchasing shares of Common Stock on the market (by using cash received through the exercise of stock options or otherwise).

Refer to Section 8.10 for application of the share limits of this Plan, including the limits in Sections 4.2 and 4.3, with respect to assumed awards. Each of the numerical limits and references in Sections 4.2 and 4.3, and in this Section 4.4, is subject to adjustment as contemplated by Section 4.3, Section 7 and Section 8.10. The share limits of Section 4.3 shall be applied on a one-for-one basis without applying the Full-Value Award premium counting rule taken into account in determining the Share Limit.

4.5 No Fractional Shares; Minimum Issue. Unless otherwise expressly provided by the Administrator, no fractional shares shall be delivered under this Plan. The Administrator may pay cash in lieu of any fractional shares in settlements of awards under this Plan. The Administrator may from time to time impose a limit (of not greater than 100 shares) on the minimum number of shares that may be purchased or exercised as to awards (or any particular award) granted under this Plan unless (as to any particular award) the total number purchased or exercised is the total number at the time available for purchase or exercise under the award.

#### 5. AWARDS

**5.1 Type and Form of Awards**. The Administrator shall determine the type or types of award(s) to be made to each selected Eligible Person. Awards may be granted singly, in combination or in tandem. Awards also may be made in combination or in tandem with, in replacement of, as alternatives to, or as the payment form for grants or rights under any other employee or compensation plan of the Corporation or one

of its Subsidiaries. The types of awards that may be granted under this Plan are:

- **5.1.1 Stock Options**. A stock option is the grant of a right to purchase a specified number of shares of Common Stock during a specified period as determined by the Administrator. An option may be intended as an incentive stock option within the meaning of Section 422 of the Code (an "ISO") or a nonqualified stock option (an option not intended to be an ISO). The agreement evidencing the grant of an option will indicate if the option is intended as an ISO; otherwise it will be deemed to be a nonqualified stock option. The maximum term of each option (ISO or nonqualified) shall be ten (10) years. The per share exercise price for each option shall be not less than 100% of the fair market value of a share of Common Stock on the date of grant of the option. When an option is exercised, the exercise price for the shares to be purchased shall be paid in full in cash or such other method permitted by the Administrator consistent with Section 5.4.
- 5.1.2 Additional Rules Applicable to ISOs. To the extent that the aggregate fair market value (determined at the time of grant of the applicable option) of stock with respect to which ISOs first become exercisable by a participant in any calendar year exceeds \$100,000, taking into account both Common Stock subject to ISOs under this Plan and stock subject to ISOs under all other plans of the Corporation or one of its Subsidiaries (or any parent or predecessor corporation to the extent required by and within the meaning of Section 422 of the Code and the regulations promulgated thereunder), such options shall be treated as nonqualified stock options. In reducing the number of options treated as ISOs to meet the \$100,000 limit, the most recently granted options shall be reduced first. To the extent a reduction of simultaneously granted options is necessary to meet the \$100,000 limit, the Administrator may, in the manner and to the extent permitted by law, designate which shares of Common Stock are to be treated as shares acquired pursuant to the exercise of an ISO. ISOs may only be granted to employees of the Corporation or one of its subsidiaries (for this purpose, the term "subsidiary" is used as defined in Section 424(f) of the Code, which generally requires an unbroken chain of ownership of at least 50% of the total combined voting power of all classes of stock of each subsidiary in the chain beginning with the Corporation and ending with the subsidiary in question). No ISO may be granted to any person who, at the time the option is granted, owns (or is deemed to own under Section 424(d) of the Code) shares of outstanding Common Stock possessing more than 10% of the total combined voting power of all classes of stock of the Corporation, unless the exercise price of such option is at least 110% of the fair market value of the stock subject to the option and such option by its terms is not exercisable after the expiration of five years from the date such option is granted. If an otherwise-intended ISO fails to meet the applicable requirements of Section 422 of the Code, the option shall be a nonqualified stock option.
- **5.1.3** *Stock Appreciation Rights.* A stock appreciation right or "SAR" is a right to receive a payment, in cash and/or Common Stock, equal to the excess of the fair market value of a specified number of shares of Common Stock on the date the SAR is exercised over the "base price" of the award, which base price shall be set forth in the applicable award agreement and shall be not less than 100% of the fair market value of a share of Common Stock on the date of grant of the SAR. The maximum term of a SAR shall be ten (10) years.

- 5.1.4 Other Awards; Dividend Equivalent Rights. The other types of awards that may be granted under this Plan include: (a) stock bonuses, restricted stock, performance stock, stock units, restricted stock units, deferred shares, phantom stock or similar rights to purchase or acquire shares, whether at a fixed or variable price (or no price) or fixed or variable ratio related to the Common Stock, and any of which may (but need not) be fully vested at grant or vest upon the passage of time, the occurrence of one or more events, the satisfaction of performance criteria or other conditions, or any combination thereof; or (b) cash awards. The types of cash awards that may be granted under this Plan include the opportunity to receive a payment for the achievement of one or more goals established by the Administrator, on such terms as the Administrator may provide, as well as discretionary cash awards. Dividend equivalent rights may be granted as a separate award or in connection with another award under this Plan; provided, however, that dividend equivalent rights may not be granted as to a stock option or SAR granted under this Plan. In addition, any dividends and/or dividend equivalents as to the portion of an award that is subject to unsatisfied vesting requirements will be subject to termination and forfeiture to the same extent as the corresponding portion of the award to which they relate in the event the applicable vesting requirements are not satisfied.
- **5.2 Award Agreements.** Each award shall be evidenced by a written or electronic award agreement or notice in a form approved by the Administrator (an "award agreement"), and, in each case and if required by the Administrator, executed or otherwise electronically accepted by the recipient of the award in such form and manner as the Administrator may require.
- **Deferrals and Settlements.** Payment of awards may be in the form of cash, Common Stock, other awards or combinations thereof as the Administrator shall determine, and with such restrictions (if any) as it may impose. The Administrator may also require or permit participants to elect to defer the issuance of shares or the settlement of awards in cash under such rules and procedures as it may establish under this Plan. The Administrator may also provide that deferred settlements include the payment or crediting of interest or other earnings on the deferral amounts, or the payment or crediting of dividend equivalents where the deferred amounts are denominated in shares.
- 5.4 Consideration for Common Stock or Awards. The purchase price (if any) for any award granted under this Plan or the Common Stock to be delivered pursuant to an award, as applicable, may be paid by means of any lawful consideration as determined by the Administrator, including, without limitation, one or a combination of the following methods:
  - (a) services rendered by the recipient of such award;
  - (b) cash, check payable to the order of the Corporation, or electronic funds transfer;
  - (c) notice and third party payment in such manner as may be authorized by the Administrator;
  - (d) the delivery of previously owned shares of Common Stock;
  - (e) by a reduction in the number of shares otherwise deliverable pursuant to the award; or
  - (f) subject to such procedures as the Administrator may adopt, pursuant to a "cashless exercise" with a third party who provides financing for the purposes of (or who otherwise facilitates) the purchase or exercise of awards.

In no event shall any shares newly-issued by the Corporation be issued for less than the minimum lawful consideration for such shares or for consideration other than consideration permitted by applicable state law. Shares of Common Stock used to satisfy the exercise price of an option shall be valued at their fair market value. The Corporation will not be obligated to deliver any shares unless and until it receives full payment of the exercise or purchase price therefor and any related withholding obligations under Section 8.5 and any other conditions to exercise or purchase have been satisfied. Unless otherwise expressly provided in the applicable award agreement, the Administrator may at any time eliminate or limit a participant's ability to pay any purchase or exercise price of any award or shares by any method other than cash payment to the Corporation.

5.5 **Definition of Fair Market Value**. For purposes of this Plan, "fair market value" shall mean, unless otherwise determined or provided by the Administrator in the circumstances, the closing price (in regular trading) for a share of Common Stock on the NASDAQ Stock Market (the "Market") for the date in question or, if no sales of Common Stock were reported on the Market on that date, the closing price (in regular trading) for a share of Common Stock on the Market for the next preceding day on which sales of Common Stock were reported on the Market. The Administrator may, however, provide with respect to one or more awards that the fair market value shall equal the closing price (in regular trading) for a share of Common Stock on the Market on the last trading day preceding the date in question or the average of the high and low trading prices of a share of Common Stock on the Market for the date in question or the most recent trading day. If the Common Stock is no longer listed or is no longer actively traded on the Market as of the applicable date, the fair market value of the Common Stock shall be the value as reasonably determined by the Administrator for purposes of the award in the circumstances. The Administrator also may adopt a different methodology for determining fair market value with respect to one or more awards if a different methodology is necessary or advisable to secure any intended favorable tax, legal or other treatment for the particular award(s) (for example, and without limitation, the Administrator may provide that fair market value for purposes of one or more awards will be based on an average of closing prices (or the average of high and low daily trading prices) for a specified period preceding the relevant date).

# 5.6 Transfer Restrictions.

- **5.6.1 Limitations on Exercise and Transfer.** Unless otherwise expressly provided in (or pursuant to) this Section 5.6 or required by applicable law: (a) all awards are non-transferable and shall not be subject in any manner to sale, transfer, anticipation, alienation, assignment, pledge, encumbrance or charge; (b) awards shall be exercised only by the participant; and (c) amounts payable or shares issuable pursuant to any award shall be delivered only to (or for the account of) the participant.
- **5.6.2** Exceptions. The Administrator may permit awards to be exercised by and paid to, or otherwise transferred to, other persons or entities pursuant to such conditions and procedures, including limitations on subsequent transfers, as the Administrator may, in its sole discretion, establish in writing. Any permitted transfer shall be subject to compliance with applicable federal and state securities laws and shall not be for value (other than nominal consideration, settlement of marital property rights, or for interests in an entity in which more than 50% of the voting interests are held by the Eligible Person or by the Eligible Person's family members).

- **5.6.3** *Further Exceptions to Limits on Transfer.* The exercise and transfer restrictions in Section 5.6.1 shall not apply to:
  - (a) transfers to the Corporation (for example, in connection with the expiration or termination of the award);
  - (b) the designation of a beneficiary to receive benefits in the event of the participant's death or, if the participant has died, transfers to or exercise by the participant's beneficiary, or, in the absence of a validly designated beneficiary, transfers by will or the laws of descent and distribution;
- (c) subject to any applicable limitations on ISOs, transfers to a family member (or former family member) pursuant to a domestic relations order if received by the Administrator;
- (d) if the participant has suffered a disability, permitted transfers or exercises on behalf of the participant by his or her legal representative; or
- (e) the authorization by the Administrator of "cashless exercise" procedures with third parties who provide financing for the purpose of (or who otherwise facilitate) the exercise of awards consistent with applicable laws and any limitations imposed by the Administrator
- 5.7 International Awards. One or more awards may be granted to Eligible Persons who provide services to the Corporation or one of its Subsidiaries outside of the United States. Any awards granted to such persons may be granted pursuant to the terms and conditions of any applicable sub-plans, if any, appended to this Plan and approved by the Administrator from time to time. The awards so granted need not comply with other specific terms of this Plan, provided that stockholder approval of any deviation from the specific terms of this Plan is not required by applicable law or any applicable listing agency.

#### 6. EFFECT OF TERMINATION OF EMPLOYMENT OR SERVICE ON AWARDS

- 6.1 General. The Administrator shall establish the effect (if any) of a termination of employment or service on the rights and benefits under each award under this Plan and in so doing may make distinctions based upon, inter alia, the cause of termination and type of award. If the participant is not an employee of the Corporation or one of its Subsidiaries, is not a member of the Board, and provides other services to the Corporation or one of its Subsidiaries, the Administrator shall be the sole judge for purposes of this Plan (unless a contract or the award otherwise provides) of whether the participant continues to render services to the Corporation or one of its Subsidiaries and the date, if any, upon which such services shall be deemed to have terminated.
- 6.2 Events Not Deemed Terminations of Employment. Unless the express policy of the Corporation or one of its Subsidiaries, or the Administrator, otherwise provides, or except as otherwise required by applicable law, the employment relationship shall not be considered terminated in the case of: (a) sick leave, (b) military leave, or (c) any other leave of absence authorized by the Corporation or one of its Subsidiaries, or the Administrator; provided that, unless reemployment upon the expiration of such leave is guaranteed by contract or law or the Administrator otherwise provides, such leave is for a period of not more than three months. In the case of any employee of the Corporation or one of its Subsidiaries on an approved leave of absence, continued vesting of the award while on leave from the employ of the Corporation or one of its Subsidiaries may be suspended until the employee returns to service, unless the Administrator otherwise provides or applicable law otherwise requires. In no event shall an award be exercised after the expiration of any applicable maximum term of the award.

**6.3 Effect of Change of Subsidiary Status.** For purposes of this Plan and any award, if an entity ceases to be a Subsidiary of the Corporation a termination of employment or service shall be deemed to have occurred with respect to each Eligible Person in respect of such Subsidiary who does not continue as an Eligible Person in respect of the Corporation or another Subsidiary that continues as such after giving effect to the transaction or other event giving rise to the change in status unless the Subsidiary that is sold, spun-off or otherwise divested (or its successor or a direct or indirect parent of such Subsidiary or successor) assumes the Eligible Person's award(s) in connection with such transaction.

#### 7. ADJUSTMENTS; ACCELERATION

#### 7.1 Adjustments.

- (a) Subject to Section 7.2, upon (or, as may be necessary to effect the adjustment, immediately prior to): any reclassification, recapitalization, stock split (including a stock split in the form of a stock dividend) or reverse stock split; any merger, combination, consolidation, conversion or other reorganization; any spin-off, split-up, or similar extraordinary dividend distribution in respect of the Common Stock; or any exchange of Common Stock or other securities of the Corporation, or any similar, unusual or extraordinary corporate transaction in respect of the Common Stock; then the Administrator shall equitably and proportionately adjust: (1) the number and type of shares of Common Stock (or other securities) that thereafter may be made the subject of awards (including the specific share limits, maximums and numbers of shares set forth elsewhere in this Plan); (2) the number, amount and type of shares of Common Stock (or other securities or property) subject to any outstanding awards; (3) the grant, purchase, or exercise price (which term includes the base price of any SAR or similar right) of any outstanding awards; and/or (4) the securities, cash or other property deliverable upon exercise or payment of any outstanding awards, in each case to the extent necessary to preserve (but not increase) the level of incentives intended by this Plan and the then-outstanding awards.
- (b) Without limiting the generality of Section 3.4, any good faith determination by the Administrator as to whether an adjustment is required in the circumstances pursuant to this Section 7.1, and the extent and nature of any such adjustment, shall be conclusive and binding on all persons.

#### 7.2 Corporate Transactions—Assumption and Termination of Awards.

(a) Upon any event in which the Corporation does not survive, or does not survive as a public company in respect of its Common Stock (including, without limitation, a dissolution, merger, combination, consolidation, conversion, exchange of securities, or other reorganization, or a sale of all or substantially all of the business, stock or assets of the Corporation, in any case in connection with which the Corporation does not survive or does not survive as a public company in respect of its Common Stock), then the Administrator may make provision for a cash payment in settlement of, or for the termination, assumption, substitution or exchange of any or all outstanding awards or the cash, securities or property deliverable to the holder of any or all outstanding awards, based

upon, to the extent relevant under the circumstances, the distribution or consideration payable to holders of the Common Stock upon or in respect of such event. Upon the occurrence of any event described in the preceding sentence in connection with which the Administrator has made provision for the award to be terminated (and the Administrator has not made a provision for the substitution, assumption, exchange or other continuation or settlement of the award): (1) unless otherwise provided in the applicable award agreement, each then-outstanding option and SAR shall become fully vested, all shares of restricted stock then outstanding shall fully vest free of restrictions, and each other award granted under this Plan that is then outstanding shall become payable to the holder of such award (with any performance goals applicable to the award in each case being deemed met, unless otherwise provided in the award agreement, at the "target" performance level); and (2) each award (including any award or portion thereof that, by its terms, does not accelerate and vest in the circumstances) shall terminate upon the related event; provided that the holder of an option or SAR shall be given reasonable advance notice of the impending termination and a reasonable opportunity to exercise his or her outstanding vested options and SARs (after giving effect to any accelerated vesting required in the circumstances) in accordance with their terms before the termination of such awards (except that in no case shall more than ten days' notice of the impending termination be required and any acceleration of vesting and any exercise of any portion of an award that is so accelerated may be made contingent upon the actual occurrence of the event).

- (b) Without limiting the preceding paragraph, in connection with any event referred to in the preceding paragraph or any change in control event defined in any applicable award agreement, the Administrator may, in its discretion, provide for the accelerated vesting of any award or awards as and to the extent determined by the Administrator in the circumstances.
- (c) For purposes of this Section 7.2, an award shall be deemed to have been "assumed" if (without limiting other circumstances in which an award is assumed) the award continues after an event referred to above in this Section 7.2, and/or is assumed and continued by the surviving entity following such event (including, without limitation, an entity that, as a result of such event, owns the Corporation or all or substantially all of the Corporation's assets directly or through one or more subsidiaries (a "Parent")), and confers the right to purchase or receive, as applicable and subject to vesting and the other terms and conditions of the award, for each share of Common Stock subject to the award immediately prior to the event, the consideration (whether cash, shares, or other securities or property) received in the event by the stockholders of the Corporation for each share of Common Stock sold or exchanged in such event (or the consideration received by a majority of the stockholders participating in such event if the stockholders were offered a choice of consideration); provided, however, that if the consideration offered for a share of Common Stock in the event is not solely the ordinary common stock of a successor corporation or a Parent, the Administrator may provide for the consideration to be received upon exercise or payment of the award, for each share subject to the award, to be solely ordinary common stock of the successor corporation or a Parent equal in fair market value to the per share consideration received by the stockholders participating in the event

- (d) The Administrator may adopt such valuation methodologies for outstanding awards as it deems reasonable in the event of a cash or property settlement and, in the case of options, SARs or similar rights, but without limitation on other methodologies, may base such settlement solely upon the excess if any of the per share amount payable upon or in respect of such event over the exercise or base price of the award. In the case of an option, SAR or similar right as to which the per share amount payable upon or in respect of such event is less than or equal to the exercise or base price of the award, the Administrator may terminate such award in connection with an event referred to in this Section 7.2 without any payment in respect of such award.
- (e) In any of the events referred to in this Section 7.2, the Administrator may take such action contemplated by this Section 7.2 prior to such event (as opposed to on the occurrence of such event) to the extent that the Administrator deems the action necessary to permit the participant to realize the benefits intended to be conveyed with respect to the underlying shares. Without limiting the generality of the foregoing, the Administrator may deem an acceleration and/or termination to occur immediately prior to the applicable event and, in such circumstances, will reinstate the original terms of the award if an event giving rise to an acceleration and/or termination does not occur.
- (f) Without limiting the generality of Section 3.4, any good faith determination by the Administrator pursuant to its authority under this Section 7.2 shall be conclusive and binding on all persons.
- (g) The Administrator may override the provisions of this Section 7.2 by express provision in the award agreement and may accord any Eligible Person a right to refuse any acceleration, whether pursuant to the award agreement or otherwise, in such circumstances as the Administrator may approve. The portion of any ISO accelerated in connection with an event referred to in this Section 7.2 (or such other circumstances as may trigger accelerated vesting of the award) shall remain exercisable as an ISO only to the extent the applicable \$100,000 limitation on ISOs is not exceeded. To the extent exceeded, the accelerated portion of the option shall be exercisable as a nonqualified stock option under the Code.
- **7.3 Definition of Change in Control**. With respect to a particular award granted under this Plan, a "**Change in Control**" shall be deemed to have occurred as of the first day, after the date of grant of the particular award, that any one or more of the following conditions shall have been satisfied:
  - (a) The acquisition by any individual, entity or group (within the meaning of Section 13(d)(3) or 14(d)(2) of the Exchange Act (a "Person")) of beneficial ownership (within the meaning of Rule 13d-3 promulgated under the Exchange Act) of 30% or more of either (1) the then-outstanding shares of Common Stock of the Corporation (the "Outstanding Corporation Common Stock") or (2) the combined voting power of the then-outstanding voting securities of the Corporation entitled to vote generally in the election of directors (the "Outstanding Corporation Voting Securities"); provided, however, that, for purposes of this definition, the following acquisitions shall not constitute a Change in Control Event; (A) any acquisition directly from the Corporation, (B) any acquisition by the Corporation, (C) any acquisition by any employee benefit plan (or related trust) sponsored or maintained by the Corporation or any affiliate of the Corporation or a successor, or (D) any acquisition by any entity pursuant to a transaction that complies with Sections (c)(1), (2) and (3) below;

- (b) Individuals who, as of the Closing (as such term is defined in Section 8.6.1), constitute the Board (the "**Incumbent Board**") cease for any reason to constitute at least a majority of the Board; provided, however, that any individual becoming a director subsequent to the Effective Date whose election, or nomination for election by the Corporation's stockholders, was approved by a vote of at least two-thirds of the directors then comprising the Incumbent Board (including for these purposes, the new members whose election or nomination was so approved, without counting the member and his predecessor twice) shall be considered as though such individual were a member of the Incumbent Board, but excluding, for this purpose, any such individual whose initial assumption of office occurs as a result of an actual or threatened election contest with respect to the election or removal of directors or other actual or threatened solicitation of proxies or consents by or on behalf of a Person other than the Board;
- Consummation of a reorganization, merger, statutory share exchange or consolidation or similar corporate transaction involving the Corporation or any of its Subsidiaries, a sale or other disposition of all or substantially all of the assets of the Corporation, or the acquisition of assets or stock of another entity by the Corporation or any of its Subsidiaries (each, a "Business Combination"), in each case unless, following such Business Combination, (1) all or substantially all of the individuals and entities that were the beneficial owners of the Outstanding Corporation Common Stock and the Outstanding Corporation Voting Securities immediately prior to such Business Combination beneficially own, directly or indirectly, more than 50% of the then-outstanding shares of common stock and the combined voting power of the then-outstanding voting securities entitled to vote generally in the election of directors, as the case may be, of the entity resulting from such Business Combination (including, without limitation, a Parent, as defined above) in substantially the same proportions as their ownership immediately prior to such Business Combination of the Outstanding Corporation Common Stock and the Outstanding Corporation Voting Securities, as the case may be, (2) no Person (excluding any entity resulting from such Business Combination or a Parent or any employee benefit plan (or related trust) of the Corporation or such entity resulting from such Business Combination or Parent) beneficially owns, directly or indirectly, 30% or more of, respectively, the then-outstanding shares of common stock of the entity resulting from such Business Combination or the combined voting power of the then-outstanding voting securities of such entity, except to the extent that the ownership in excess of 30% existed prior to the Business Combination, and (3) at least a majority of the members of the board of directors or trustees of the entity resulting from such Business Combination or a Parent were members of the Incumbent Board at the time of the execution of the initial agreement or of the action of the Board providing for such Business Combination; or
- (d) Approval by the stockholders of the Corporation of a complete liquidation or dissolution of the Corporation other than in the context of a transaction that does not constitute a Change in Control Event under clause (c) above.

#### 8. OTHER PROVISIONS

- 8.1 Compliance with Laws. This Plan, the granting and vesting of awards under this Plan, the offer, issuance and delivery of shares of Common Stock, and/or the payment of money under this Plan or under awards are subject to compliance with all applicable federal, state, local and foreign laws, rules and regulations (including, but not limited to, state and federal securities law and federal margin requirements) and to such approvals by any listing, regulatory or governmental authority as may, in the opinion of counsel for the Corporation, be necessary or advisable in connection therewith. The person acquiring any securities under this Plan will, if requested by the Corporation or one of its Subsidiaries, provide such assurances and representations to the Corporation or one of its Subsidiaries as the Administrator may deem necessary or desirable to assure compliance with all applicable legal and accounting requirements.
- **8.2 No Rights to Award.** No person shall have any claim or rights to be granted an award (or additional awards, as the case may be) under this Plan, subject to any express contractual rights (set forth in a document other than this Plan) to the contrary.
- 8.3 No Employment/Service Contract. Nothing contained in this Plan (or in any other documents under this Plan or in any award) shall confer upon any Eligible Person or other participant any right to continue in the employ or other service of the Corporation or one of its Subsidiaries, constitute any contract or agreement of employment or other service or affect an employee's status as an employee at will, nor shall interfere in any way with the right of the Corporation or one of its Subsidiaries to change a person's compensation or other benefits, or to terminate his or her employment or other service, with or without cause. Nothing in this Section 8.3, however, is intended to adversely affect any express independent right of such person under a separate employment or service contract other than an award agreement.
- 8.4 Plan Not Funded. Awards payable under this Plan shall be payable in shares or from the general assets of the Corporation, and no special or separate reserve, fund or deposit shall be made to assure payment of such awards. No participant, beneficiary or other person shall have any right, title or interest in any fund or in any specific asset (including shares of Common Stock, except as expressly otherwise provided) of the Corporation or one of its Subsidiaries by reason of any award hereunder. Neither the provisions of this Plan (or of any related documents), nor the creation or adoption of this Plan, nor any action taken pursuant to the provisions of this Plan shall create, or be construed to create, a trust of any kind or a fiduciary relationship between the Corporation or one of its Subsidiaries and any participant, beneficiary or other person. To the extent that a participant, beneficiary or other person acquires a right to receive payment pursuant to any award hereunder, such right shall be no greater than the right of any unsecured general creditor of the Corporation.
- **8.5 Tax Withholding.** Upon any exercise, vesting, or payment of any award, or upon the disposition of shares of Common Stock acquired pursuant to the exercise of an ISO prior to satisfaction of the holding period requirements of Section 422 of the Code, or upon any other tax withholding event with respect to any award, arrangements satisfactory to the Corporation shall be made to provide for any taxes the Corporation or any of its Subsidiaries may be required or permitted to withhold with respect to such award event or payment. Such

arrangements may include (but are not limited to) any one of (or a combination of) the following:

- (a) The Corporation or one of its Subsidiaries shall have the right to require the participant (or the participant's personal representative or beneficiary, as the case may be) to pay or provide for payment of the amount of any taxes which the Corporation or one of its Subsidiaries may be required or permitted to withhold with respect to such award event or payment.
- (b) The Corporation or one of its Subsidiaries shall have the right to deduct from any amount otherwise payable in cash (whether related to the award or otherwise) to the participant (or the participant's personal representative or beneficiary, as the case may be) the amount of any taxes which the Corporation or one of its Subsidiaries may be required or permitted to withhold with respect to such award event or payment.
- (c) In any case where a tax is required to be withheld in connection with the delivery of shares of Common Stock under this Plan, the Administrator may in its sole discretion (subject to Section 8.1) require or grant (either at the time of the award or thereafter) to the participant the right to elect, pursuant to such rules and subject to such conditions as the Administrator may establish, that the Corporation reduce the number of shares to be delivered by (or otherwise reacquire) the appropriate number of shares, valued in a consistent manner at their fair market value or at the sales price in accordance with authorized procedures for cashless exercises, necessary to satisfy any applicable withholding obligation on exercise, vesting or payment.

#### 8.6 Effective Date, Termination and Suspension, Amendments.

**8.6.1 Effective Date**. This Plan was effective as of March 28, 2018, the date of its approval by the board of directors of Progenics (the "Effective Date"). Unless earlier terminated by the Board and subject to any extension that may be approved by stockholders, this Plan shall terminate at the close of business on the day before the tenth anniversary of the Effective Date. After the termination of this Plan either upon such stated termination date or its earlier termination by the Board, no additional awards may be granted under this Plan, but previously granted awards (and the authority of the Administrator with respect thereto, including the authority to amend such awards) shall remain outstanding in accordance with their applicable terms and conditions and the terms and conditions of this Plan. The Plan, as amended in connection with the Merger, became effective following its adoption by the Board and the closing of the Merger on June 19, 2020 (the "Closing").

**8.6.2 Board Authorization**. The Board may, at any time, terminate or, from time to time, amend, modify or suspend this Plan, in whole or in part. No awards may be granted during any period that the Board suspends this Plan.

**8.6.3** Stockholder Approval. To the extent then required by applicable law or deemed necessary or advisable by the Board, any amendment to this Plan shall be subject to stockholder approval.

- **8.6.4 Amendments to Awards**. Without limiting any other express authority of the Administrator under (but subject to) the express limits of this Plan, the Administrator by agreement or resolution may waive conditions of or limitations on awards to participants that the Administrator in the prior exercise of its discretion has imposed, without the consent of a participant, and (subject to the requirements of Sections 3.2 and 8.6.5) may make other changes to the terms and conditions of awards. Any amendment or other action that would constitute a repricing of an award is subject to the no-repricing provision of Section 3.3.
- **8.6.5** *Limitations on Amendments to Plan and Awards*. No amendment, suspension or termination of this Plan or amendment of any outstanding award agreement shall, without written consent of the participant, affect in any manner materially adverse to the participant any rights or benefits of the participant or obligations of the Corporation under any award granted under this Plan prior to the effective date of such changes, settlements and other actions contemplated by Section 7 shall not be deemed to constitute changes or amendments for purposes of this Section 8.6.
- **8.7 Privileges of Stock Ownership.** Except as otherwise expressly authorized by the Administrator, a participant shall not be entitled to any privilege of stock ownership as to any shares of Common Stock not actually delivered to and held of record by the participant. Except as expressly required by Section 7.1 or otherwise expressly provided by the Administrator, no adjustment will be made for dividends or other rights as a stockholder for which a record date is prior to such date of delivery.
- 8.8 Governing Law; Severability.
  - **8.8.1 Choice of Law**. This Plan, the awards, all documents evidencing awards and all other related documents shall be governed by, and construed in accordance with the laws of the State of Delaware, notwithstanding any Delaware or other conflict of law provision to the contrary.
  - **8.8.2** Severability. If a court of competent jurisdiction holds any provision invalid and unenforceable, the remaining provisions of this Plan shall continue in effect.
- **8.9 Captions.** Captions and headings are given to the sections and subsections of this Plan solely as a convenience to facilitate reference. Such headings shall not be deemed in any way material or relevant to the construction or interpretation of this Plan or any provision thereof.
- 8.10 Stock-Based Awards in Substitution for Stock Options or Awards Granted by Other Corporation. Awards may be granted to Eligible Persons in substitution for or in connection with an assumption of employee stock options, SARs, restricted stock or other stock-based awards granted by other entities to persons who are or who will become Eligible Persons in respect of the Corporation or one of its Subsidiaries, in connection with a distribution, merger or other reorganization by or with the granting entity or an affiliated entity, or the acquisition by the Corporation or one of its Subsidiaries, directly or indirectly, of all or a substantial part of the stock or assets of the employing entity. The awards so granted need not comply with other specific terms of this Plan, provided the awards reflect adjustments giving effect to the assumption or substitution consistent with any conversion applicable to the common stock (or the securities otherwise subject to the award) in the transaction and any change in the issuer of the security. Any shares that are delivered and any awards that are granted by, or become obligations of, the Corporation, as a result of the assumption by the Corporation of, or in substitution for, outstanding awards previously granted or assumed by an acquired company (or previously granted or assumed by a predecessor employer (or direct or indirect parent thereof) in the case of persons that become employed by the Corporation or one of its Subsidiaries in connection with a business or asset acquisition or similar transaction) shall not be counted against the Share Limit or other limits on the number of shares available for issuance under this Plan.

- **8.11 Non-Exclusivity of Plan.** Nothing in this Plan shall limit or be deemed to limit the authority of the Board or the Administrator to grant awards or authorize any other compensation, with or without reference to the Common Stock, under any other plan or authority.
- 8.12 No Corporate Action Restriction. The existence of this Plan, the award agreements and the awards granted hereunder shall not limit, affect, or restrict in any way the right or power of the Corporation or any Subsidiary (or any of their respective shareholders, boards of directors or committees thereof (or any subcommittees), as the case may be) to make or authorize: (a) any adjustment, recapitalization, reorganization or other change in the capital structure or business of the Corporation or any Subsidiary, (b) any merger, amalgamation, consolidation or change in the ownership of the Corporation or any Subsidiary, (c) any issue of bonds, debentures, capital, preferred or prior preference stock ahead of or affecting the capital stock (or the rights thereof) of the Corporation or any Subsidiary, (d) any dissolution or liquidation of the Corporation or any Subsidiary, (e) any sale or transfer of all or any part of the assets or business of the Corporation or any Subsidiary, (f) any other award, grant, or payment of incentives or other compensation under any other plan or authority (or any other action with respect to any benefit, incentive or compensation), or (g) any other corporate act or proceeding by the Corporation or any Subsidiary. No participant, beneficiary or any other person shall have any claim under any award or award agreement against any member of the Board or the Administrator, or the Corporation or any employees, officers or agents of the Corporation or any Subsidiary, as a result of any such action. Awards need not be structured so as to be deductible for tax purposes.
- 8.13 Other Company Benefit and Compensation Programs. Payments and other benefits received by a participant under an award made pursuant to this Plan shall not be deemed a part of a participant's compensation for purposes of the determination of benefits under any other employee welfare or benefit plans or arrangements, if any, provided by the Corporation or any Subsidiary, except where the Administrator expressly otherwise provides or authorizes in writing. Awards under this Plan may be made in addition to, in combination with, as alternatives to or in payment of grants, awards or commitments under any other plans, arrangements or authority of the Corporation or its Subsidiaries.
- **8.14 Clawback Policy**. The awards granted under this Plan are subject to the terms of the Corporation's recoupment, clawback or similar policy as it may be in effect from time to time, as well as any similar provisions of applicable law, any of which could in certain circumstances require repayment or forfeiture of awards or any shares of Common Stock or other cash or property received with respect to the awards (including any value received from a disposition of the shares acquired upon payment of the awards).

# WHITE & CASE

White & Case LLP 1221 Avenue of the Americas New York, NY 10020-1095 T+1 212 819 8200

whitecase.com

June 26, 2020

Lantheus Holdings, Inc. 331 Treble Cove Road North Billerica, Massachusetts 01862

Re: Registration Statement on Form S-8

#### Ladies and Gentlemen:

We have acted as counsel to Lantheus Holdings, Inc., a corporation organized under the laws of Delaware ("<u>Lantheus Holdings</u>"), in connection with the preparation and filing with the Securities and Exchange Commission (the "<u>SEC</u>") under the Securities Act of 1933 (the "<u>Securities Act</u>") of Lantheus Holdings' Registration Statement on Form S-8 (the "<u>Registration Statement</u>," which term does not include any other document or agreement whether or not specifically referred to or incorporated by reference therein or attached as an exhibit or schedule thereto), relating to up to 4,211,290 shares of Lantheus Holdings' common stock, par value \$0.01 per share (the "<u>Lantheus Holdings Shares</u>"), to be issued by Lantheus Holdings from time to time to holders of certain options granted under the Lantheus Holdings, Inc. 2018 Performance Incentive Plan (f/k/a Progenics Pharmaceuticals, Inc. 2018 Performance Incentive Plan), as amended, and the Lantheus Holdings, Inc. 2005 Stock Incentive Plan (f/k/a Progenics Pharmaceuticals, Inc. 2005 Stock Incentive Plan), as amended (collectively, the "<u>Plans</u>"), which were converted pursuant to that certain Amended and Restated Agreement and Plan of Merger (the "<u>Merger Agreement</u>"), dated as of February 20, 2020, by and among Lantheus Holdings, Plato Merger Sub, Inc., a Delaware corporation and a wholly owned subsidiary of Lantheus Holdings, and Progenics Pharmaceuticals, Inc., a Delaware corporation.

In connection with our opinions expressed below, we have examined originals or copies certified to our satisfaction of the following documents and such other documents, certificates and other statements of government officials and corporate officers of Lantheus Holdings as we deemed necessary for the purposes of the opinions set forth in this opinion letter: (i) the Registration Statement in the form to be filed with the SEC on the date hereof, (ii) the Merger Agreement, (iii) the certificate of merger filed with the Secretary of State of Delaware in connection with the Merger Agreement, (iv) the Plans, (v) Lantheus Holdings' Amended and Restated Bylaws and (vii) a form of the share certificate, which has been incorporated by reference as an exhibit to the Registration Statement. We have also examined the originals, or duplicative or certificated or conformed copies, of such corporate records, agreements, documents and other instruments and have made such other investigations as we have deemed relevant and necessary in connection with this opinion letter. As to questions of fact material to this opinion letter, we have relied, with your approval, upon oral and written representations of Lantheus Holdings and certificates or comparable documents of public officials and of officers and representatives of Lantheus Holdings.



In making such examination and rendering such opinions, we have assumed without independent investigation or verification of any kind the genuineness of all signatures, the legal capacity of all natural persons signing all documents, the authenticity of all documents submitted to us as originals, the conformity to authentic original documents of all documents submitted to us as copies, the truthfulness, completeness and correctness of all factual representations and statements contained in all documents, and the accuracy and completeness of all public records examined by us.

Based upon the foregoing assumptions, and subject to the qualifications set forth in this opinion letter, having considered such questions of law as we have deemed necessary as a basis for the opinion expressed below, we are of the opinion that the Lantheus Holdings Shares to be issued by Lantheus Holdings pursuant to and in the manner contemplated by the terms of the Merger Agreement and the Plans and the award agreements thereunder will be, upon issuance, duly authorized and, when the Lantheus Holdings Shares have been issued in accordance with the terms and conditions set forth in the Registration Statement, the Merger Agreement and the Plans and the award agreements thereunder, such Lantheus Holdings Shares will be validly issued, fully paid and nonassessable.

We are members of the bar of the State of New York. Lantheus Holdings is a Delaware corporation, and we have not considered, and we express no opinion as to, any law other than the Delaware General Corporation Law.

We hereby consent to be named in the Registration Statement as the attorneys who passed upon the legality of the Lantheus Holdings Shares to be issued pursuant to the Registration Statement and to the filing of a copy of this opinion letter as Exhibit 5.1 to the Registration Statement. In giving such consent, we do not hereby admit that we are in the category of persons whose consent is required under Section 7 of the Securities Act, as amended, and the rules and regulations of the SEC promulgated thereunder. The opinions expressed above are as of the date hereof only, and we assume no responsibility to update this opinion letter, or to advise you of, any such facts or circumstances of which we become aware, regardless of whether or not they affect the opinions expressed in this opinion letter.

Very truly yours,

/s/ White & Case LLP

# CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We consent to the incorporation by reference in this Registration Statement on Form S-8 of our reports dated February 25, 2020, relating to the financial statements of Lantheus Holdings, Inc. and the effectiveness of Lantheus Holdings, Inc.'s internal control over financial reporting, appearing in the Annual Report on Form 10-K of Lantheus Holdings, Inc. for the year ended December 31, 2019.

/s/ Deloitte & Touche LLP

Boston, Massachusetts June 26, 2020

# Consent of Independent Registered Public Accounting Firm

We consent to the incorporation by reference in the Registration Statement (Form S-8) pertaining to the Lantheus Holdings, Inc. 2018 Performance Incentive Plan, as amended, and the Lantheus Holdings, Inc. 2005 Stock Incentive Plan, as amended, of our report dated March 13, 2020, with respect to the consolidated financial statements of Progenics Pharmaceuticals, Inc. for the years ended December 31, 2019 and 2018 appearing in Lantheus Holdings Inc.'s Current Report on Form 8-K, filed with the Securities and Exchange Commission on June 22, 2020.

/s/ Ernst & Young LLP

Stamford, Connecticut

June 26, 2020